

Temi di discussione

(Working Papers)

Drivers of large recessions and monetary policy responses

by Giovanni Melina and Stefania Villa







Temi di discussione

(Working Papers)

Drivers of large recessions and monetary policy responses

by Giovanni Melina and Stefania Villa

Number 1425 - October 2023

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Antonio Di Cesare, Raffaela Giordano, Marco Bottone, Lorenzo Braccini, Mario Cannella, Giacomo Caracciolo, Rosalia Greco, Valerio Della Corte, Lucia Esposito, Danilo Liberati, Salvatore Lo Bello, Alessandro Moro, Claudia Pacella, Marco Pelosi, Fabio Piersanti, Dario Ruzzi, Stefania Villa. *Editorial Assistants:* Roberto Marano.

ISSN 2281-3950 (online)

Designed by the Printing and Publishing Division of the Bank of Italy

DRIVERS OF LARGE RECESSIONS AND MONETARY POLICY RESPONSES

by Giovanni Melina* and Stefania Villa**

Abstract

Shocks to capital utilization are introduced into a structural macroeconomic closedeconomy model with financial frictions to capture disruptions to the ability of the capital stock to provide capital services used in production. Estimates for the euro area and the United States show that these shocks were among the most important drivers of the output contraction during the Global Financial Crisis and the COVID-19 crisis, while financial shocks were more significant only during the Global Financial Crisis. Thanks to the timely and strong intervention of the European Central Bank and the US Federal Reserve, monetary policy shocks exerted a sizeable and positive contribution to output and inflation during the COVID-19 crisis.

JEL Classification: E4, E5, E6.

Keywords: COVID-19, Global Financial Crisis, Great Lockdown, monetary policy, capital utilization.

DOI: 10.32057/0.TD.2023.1425

Contents

1.	Introduction	5
2.	Model	8
3.	Estimation	12
	3.1 Data and measurement equations	12
	3.2 Calibration and estimates	13
4.	Results	17
	4.1 Dynamic properties of the estimated model	18
	4.2 The role of shocks on economic activity and inflation	23
	4.3 The contribution of monetary policy	27
5.	Conclusions	30
Re	ferences	31
Aŗ	ppendix A - Data	37
Aŗ	ppendix B - Model Dynamics	37
Aŗ	ppendix C - Robustness checks	39

^{*} International Monetary Fund.

^{**} Bank of Italy, Directorate General Economics, Statistics and Research.

1 Introduction¹

The global economy and much of the macroeconomic research have been dominated by the two large recessions occurred in the past fifteen years: the Great Recession that followed the Global Financial Crisis (GFC) and the COVID-19 crisis (C-19C) that followed the pandemic. Policymakers across the world came up with creative solutions to tackle the challenges created by these major crises. For instance, the GFC prompted central banks in much of the industrialized world to take unconventional monetary policy actions to overcome the constraint posed by a binding zero lower bound (ZLB) on the policy rate. COVID-19 also posed economic policy challenges, and arguably policymakers could rely on the experience acquired during the GFC to take swift policy decisions. Countries implemented, among others, packages of fiscal measures, such as discretionary fiscal stimulus measures, state guarantees for loans to firms and other liquidity support measures in response to the pandemic. Overall, the experience gained in these two crises may prove useful in the future to counter the adverse effects of rare and large shocks.

While sharing some similarities, such as an abrupt fall in output and the increase in uncertainty, these crises were originated by different shocks: financial shocks in the case of the GFC and a pandemic in that of the C-19C. Understanding the impact of these shocks on macroeconomic fluctuations is a precondition for calibrating the policy tools. For example, easing monetary policy in a demand-driven recession such as the GFC,² characterized by a negative output gap and deflationary pressures, is an uncontroversial policy action. Choosing the appropriate monetary policy stance in an environment such as that created by the COVID-19 shock with a combination of supply and demand factors is a much more difficult endeavor.

Focusing on two of the world's largest economies, the Euro Area (EA) and the United States, this paper attempts to answer two research questions: (i) which shocks drove the output contraction and inflation fluctuations during the GFC and C-19C? And (ii) to what extent monetary policy contributed to the recovery from these major crises?

These questions are tackled with the Bayesian estimation of a Dynamic Stochastic General Equilibrium (DSGE) model \dot{a} la Smets and Wouters (2007) augmented with financial frictions in the spirit of Gertler and Karadi (2011). The model features a novel exogenous disturbance to capital utilization to capture disruptions that impair the ability of the capital stock to provide capital services used in production. As shown by Santacreu (2016), capacity utilization usually falls sharply during recessions as firms adjust their

¹The views expressed in this paper are those of the authors and do not necessarily represent those of the International Monetary Fund or IMF policy, or those of the Bank of Italy. We are grateful to Stefano Neri, Tiziano Ropele, Giodano Zevi, Roberta Zizza, seminar participants at Banca d'Italia and many other colleagues for very useful comments. All remaining errors are ours.

²It should be noted that there is a debate in the literature on whether financial shocks act as a demand or a supply shock. This paper finds that it behaves as a demand shock in line with several contributions (e.g. Gerali et al., 2010; Furlanetto et al., 2019, among others).

productive capacity to face lower demand, as in the case of the GFC. In the case of the C-19C, capacity was heavily underutilized due to lockdowns and other containment measures. The model is estimated with macroeconomic and financial variables using data of the EA and the US economy. In both cases, estimates show that the absence of a capital utilization shock would lead to a severe underestimation of the drop in capacity utilization observed in the data during the GFC and the C-19C.

The sample covers the period 2000Q1-2022Q1 and includes times during which the policy rate reached the ZLB and unconventional monetary policy measures were taken, posing potential issues in the estimation of a log-linearized model. Following Mouabbi and Sahuc (2019) and Batini et al. (2021), the policy rate is thus replaced with the shadow rate constructed by Wu and Xia (2016, 2017, 2020).³ This rate is a synthetic policy rate that takes into account the effects of all the conventional and unconventional monetary policies implemented by the central banks. In addition, it is free to move in the negative territory, thus circumventing the estimation issues posed by the ZLB.

The main findings can be summarized as follows. First, capital utilization shocks turned out to be important contributors to the negative components of output growth, but not of inflation fluctuations, during the GFC and the C-19C in both the EA and the US economy. Second, financial shocks played a more important role during the GFC, weighing on GDP growth for about one year after the trough, and to a greater extent in the EA compared to the US economy. Third, labor and preference shocks explained a large fraction of the remaining output contraction and of inflation fluctuations in the US economy during the C-19C, while in the EA total factor productivity (TFP) and price mark-up shocks played a more prominent role. Lastly, thanks to the timely and strong intervention of the European Central Bank and the U.S. Federal Reserve, monetary policy shocks had a sizable positive contribution to the recovery from the C-19C and to movements in inflation.

There are three caveats associated with this analysis. First, having focused on the role of monetary policy, the various fiscal measures adopted during the crisis periods have not been modeled explicitly. Moreover, the recovery after the C-19C has been influenced also by the vaccination campaign, which is only indirectly captured by the softening and reversal of shocks capturing lockdown and containment measures. Finally, the model does not distinguish between essential versus non-essential services, which were affected differently during the C-19C, but captures overall effects on aggregate macroeconomic variables.⁴

The paper is related to three main of strands of the DSGE literature. The first group of papers is represented by those aiming to analyze the role of shocks during

 $^{^3{\}rm More}$ recent data on the shadow rate are available at Cynthia Wu's website, https://sites.google.com/view/jingcynthiawu/shadow-rates.

⁴The model also does not consider durable *versus* nondurable goods, although their dynamics differed particularly during the C-19C.

the COVID-19 pandemic. For instance, Kollmann (2021) examines the macroeconomic effects of the COVID-19 pandemic on EA GDP and inflation, using a stylized New Keynesian model, where this shock is interpreted as a combination of aggregate demand and aggregate supply disturbances, as in Bartocci et al. (2020). This analysis concludes that the dominant force driving the sharp fall of EA GDP in 2020 was an aggregate supply contraction and that COVID-19-induced aggregate demand and supply shifts were persistent. Cardani et al. (2022) and Corrado et al. (2021) also investigate the role of shocks during the pandemic. Similarly to the present contribution, these papers employ purely macroeconomic models where the pandemic is captured via shocks. In parallel, a related literature was developed to allow for the interaction between economic decisions and rates of infections (e.g., Kaplan et al., 2020; Eichenbaum et al., 2021; Giagheddu and Papetti, 2023, among others). These contributions do not employ traditional DSGE models, but they extend canonical epidemiological models to embed macroeconomic variables. The second strand consists of the plethora of model-based analyses on the GFC, which introduced financial frictions in otherwise standard DSGE models (Gerali et al., 2010; Kollmann et al., 2013; Kiley and Sim, 2014, among others). This paper uses banking sector frictions modeled in the form of incentive-compatibility constraints linking banks balance sheets with terminal wealth as in Gertler and Karadi (2011) because, as noted by Villa (2016), this model is effective at replicating the propagation of shocks. Most of these contributions, like this paper, do not focus on the interaction between fiscal policy and the financial sector, a channel investigated by other studies (e.g. Melina and Villa, 2014; Batini et al., 2019; Silva, 2021).⁵ The third strand of the literature is represented by those studies finding an important role for shocks to the capital accumulation process to explain business cycle fluctuations (see Gertler and Karadi, 2011; Furlanetto and Seneca, 2014; Riggi, 2019; Bottone et al., 2021, among others).

In sum, previous contributions analyzed either one recession or one country at a time, and only in a few cases they compared the monetary policy response across the two recessions (e.g. Cortes et al., 2022). This is the first study that (i) assesses the drivers of two very different recessions, the GFC and the C-19C, combined with the role of monetary policy, using the same modeling and estimation approach across the EA and the US economy;⁶; (ii) establishes the important role of capital utilization shocks in large recessions; and (iii) quantifies the macroeconomic effects of monetary policy in the EA and the US economy in relevant periods of the recent history.

The remainder of the paper is structured as follows. Section 2 outlines the DSGE

⁵For instance Silva (2021) shows that, when government spending (or government deficits) are high, the aggregate credit risk of the financial sector and the provision of financial services are compromised.

⁶The Euro Area experienced also the sovereign debt crisis, the analysis of which would have required another modelling device as in Batini et al. (2021). It should be noted that the focus of the paper is on large recessions which affected both economies. Therefore, we deem it appropriate to use the same structural macroeconomic model for the sake of comparison.

model. Section 3 presents the data and the estimation strategy. Section 4 reports the results. Finally, Section 5 concludes. Details about the data and additional results are reported in the Appendix.

2 Model

This section summarizes the DSGE model, which is a standard Smets and Wouters (2007) economy augmented with financial frictions as in Gertler and Karadi (2011), similarly to Villa (2016). As common in the literature, the economy is populated by the following agents: households, labor unions, labor packers, retailers, final good firms, intermediate goods firms, capital producers, financial intermediaries and the policymaker.

Households consume, accumulate government bonds and supply labor. A labor union differentiates labor and sets wages in a monopolistically competitive market. Competitive labor packers buy labor services from the union, and package and sell them to intermediate goods firms. The presence of an agency problem limits the ability of financial intermediaries to obtain deposits from households as in Gertler and Karadi (2011). This feature, in turn, affects the leverage ratio of financial intermediaries.⁷ Output is produced in several steps, involving a monopolistically competitive market with producers facing price rigidities. The monetary authority sets the short-term nominal interest rate according to a Taylor rule.

The remainder of this section reports the model equilibrium conditions in log-linear form in four groups: (i) households, (ii) non-financial firms, (iii) financial intermediaries, (iv) monetary authority and equilibrium.⁸

(i) Households

$$\hat{mu}_t = \frac{1}{1-h} \left(h \hat{C}_{t-1} - \hat{C}_t \right) + \varepsilon_t^b \tag{1}$$

$$\hat{\Lambda}_{t,t+1} \equiv m\hat{u}_{t+1} - \hat{mu}_t \tag{2}$$

$$\hat{\Lambda}_{t,t+1} + \hat{R}_t = 0 \tag{3}$$

$$\hat{R}_t^n = \hat{R}_t + E\left[\hat{\Pi}_{t+1}\right] \tag{4}$$

⁷As shown in Villa (2016), the introduction of financial frictions improves the models' fit compared to a standard Smets and Wouters (2007) economy, suggesting that these frictions are empirically relevant both in the EA and the US.

⁸Variables with a 'hat' denote a percentage deviation from steady state, while a variable without a time subscript denotes its steady-state value.

$$\hat{W}_{t} = \frac{\beta}{(1+\beta)} E_{t} \left[\hat{W}_{t+1} \right] + \frac{1}{(1+\beta)} \hat{W}_{t-1} + \frac{\beta}{(1+\beta)} E_{t} \left[\hat{\Pi}_{t+1} \right] - \frac{(1+\beta\sigma_{wi})}{(1+\beta)} \hat{\Pi}_{t} + \frac{\sigma_{wi}}{(1+\beta)} \hat{\Pi}_{t-1} + \frac{1}{(1+\beta)} \frac{(1-\beta\sigma_{w})(1-\sigma_{w})}{(1+\varepsilon_{w}\phi)\sigma_{w}} \times \left[\phi \hat{L}_{t} - \frac{h}{1-h} \hat{C}_{t-1} + \frac{1}{1-h} \hat{C}_{t} - \hat{W}_{t} \right] + \varepsilon_{t}^{w}$$
(5)

Equation (1) is the marginal utility of consumption, \hat{mu}_t , where h measures the degree of habit formation in consumption, \hat{C}_t , and ϵ_t^b is a preference shock. Equation (2) is the definition of the stochastic discount factor, $\hat{\Lambda}_{t,t+1}$. Equation (3) is the Euler consumption equation linking the stochastic discount factor to the real interest rate \hat{R}_t . Equation (4) represents the Fisher identity linking the nominal interest rate, \hat{R}_t^n , to the real interest rate and inflation, $\hat{\Pi}_t$. Finally, Equation (5) represents the Calvo staggered wage setting, in which \hat{W}_t is the nominal wage, \hat{L}_t is labor in terms of hours worked, β is the households' discount factor, σ_w measures wage stickiness, σ_{wi} denotes the degree of wage indexation and ϵ_t^w is a labor supply (or wage mark-up) shock. The wage mark-up, defined as $\phi \hat{L}_t - \frac{\hbar}{1-h} \hat{C}_{t-1} + \frac{1}{1-h} \hat{C}_t - \hat{W}_t$, is determined by the difference between the marginal rate of substitution between working and consuming and the real wage, with ϕ being the inverse of the Frisch elasticity of labor supply.

(ii) Non-Financial Firms

$$\hat{Y}_t = \varepsilon_t^a + \alpha \left(\varepsilon_t^k + \hat{K}_t + \hat{U}_t \right) + (1 - \alpha) \hat{L}_t$$
(6)

$$\hat{Z}_t^k = \frac{\zeta}{1-\zeta} \left(\hat{U}_t + \varepsilon_t^u \right) \tag{7}$$

$$\hat{\Pi}_{t} = \frac{\sigma_{pi}}{1 + \sigma_{pi}\beta} \hat{\Pi}_{t-1} + \frac{\beta}{1 + \sigma_{pi}\beta} E_{t} \left[\hat{\Pi}_{t+1} \right] - \frac{(1 - \beta\sigma_{p})(1 - \sigma_{p})}{(1 + \sigma_{pi}\beta)\sigma_{p}} \times \left[\varepsilon_{t}^{a} - \alpha \hat{Z}_{t}^{k} - (1 - \alpha) \hat{W}_{t} \right] + \varepsilon_{t}^{p}$$

$$\tag{8}$$

$$\hat{W}_{t} = \hat{Z}_{t}^{k} - \hat{L}_{t} + \hat{K}_{t} + \hat{U}_{t}$$
(9)

$$\hat{I}_t = \frac{1}{\xi(1+\beta)} \left(\hat{Q}_t + \varepsilon_t^x \right) + \frac{1}{(1+\beta)} \hat{I}_{t-1} + \frac{\beta}{(1+\beta)} E_t \left[\hat{I}_{t+1} \right]$$
(10)

$$\hat{R}_t^k = \frac{Z^k}{R^k} \hat{Z}_t^k + \frac{(1-\delta)}{R^k} \left(\hat{Q}_t + \varepsilon_t^k \right) - \hat{Q}_{t-1}$$
(11)

$$\hat{K}_{t+1} = \delta(\hat{I}_t + \varepsilon_t^x) + (1 - \delta) \left(\hat{K}_t + \varepsilon_t^k\right)$$
(12)

Equation (6) captures the production technology, where \hat{Y}_t is output, ϵ_t^a is a technology shock, α is the capital share and ϵ_t^k is a capital quality shock. Capital, \hat{K}_t is augmented

by the capital utilization rate, \hat{U}_t , the optimality condition of which is given by equation (7), where ζ represents the positive function of the elasticity of the capital utilization adjustment cost, \hat{Z}_t^k is the marginal product of capital and ϵ_t^u is a capital utilization shock. This shock is meant to capture exogenous disturbances affecting the process through which the capital stock turns into capital services via its utilization rate. Staggered price stickiness is incorporated in the model through limiting the ability of firms to reset their prices every period with a probability equal to σ_p , as shown by equation (8), where σ_{pi} governs the degree of price indexation and ϵ_t^p is a price mark-up shock. Cost minimization by firms leads to equation (9), implying that the marginal product of capital is negatively related to the capital-labor ratio and positively related to real wages. Investment, \hat{I}_t is described by equation (10), where \hat{Q}_t is the current value of capital stock, ξ is the elasticity of the investment adjustment cost and ϵ_t^x is an investment-specific technology shock. The arbitrage condition for the value of capital is given by equation (11), where \hat{R}_t^k is the external cost of funding and δ is the capital depreciation rate. The law of motion of installed capital is given by equation (12).

(iii) Financial Intermediaries

$$\hat{EP}_t = E_t \left[\hat{R}_{t+1}^k \right] - \hat{R}_t \tag{13}$$

$$\hat{K}_{t+1} + \hat{Q}_t = \hat{lev}_t + \hat{N}_t \tag{14}$$

$$\hat{lev}_t = \hat{D}_t + \frac{V}{\lambda - V}\hat{V}_t \tag{15}$$

$$\hat{D}_{t} = \theta \beta Z E_{t} [\hat{\Lambda}_{t,t+1} + \hat{Z}_{t,t+1} + \hat{D}_{t+1}]$$
(16)

$$\hat{Z}_{t,t+1} = \frac{1}{Z} \left[lev R^k E_t \left[\hat{R}_{t+1}^k \right] + R(1 - lev) \hat{R}_t + (R^k - R) lev \hat{lev}_t \right]$$
(17)

$$\hat{V}_{t} = \frac{(1-\theta)\beta}{V} [R^{k} - R] E_{t} \left[\hat{\Lambda}_{t,t+1} \right] + \frac{(1-\theta)\beta}{V} \left[R^{k} E_{t} \left[\hat{R}_{t+1}^{k} \right] - R\hat{R}_{t} \right] + \theta\beta X E_{t} [\hat{X}_{t,t+1} + \hat{V}_{t+1} + \hat{\Lambda}_{t,t+1}]$$
(18)

$$\hat{X}_{t,t+1} = E_t \left[\hat{lev}_{t+1} \right] + \hat{Z}_{t,t+1} - \hat{lev}_t$$
(19)

$$\hat{N}_t = \frac{N^e}{Y} \frac{Y}{N} \hat{N}_t^e + \frac{N^n}{Y} \frac{Y}{N} \hat{N}_t^n \tag{20}$$

$$\hat{N}_{t}^{e} = \hat{N}_{t-1} + \frac{1}{Z} \left[lev R^{k} E_{t} \left[\hat{R}_{t+1}^{k} \right] + R(1 - lev) \hat{R}_{t} + (R^{k} - R) lev \hat{lev}_{t} \right]$$
(21)

$$\hat{N}_t^n = \hat{Q}_t + \hat{K}_t \tag{22}$$

Financial intermediaries raise funds from households and grant loans to intermediate producers. Due to a moral-hazard costly enforcement problem, the presence of financial intermediation leads to an endogenous credit spread, \hat{EP}_t captured by equation (13), as a difference between the cost of funding state-contingent asset of non-financial firms and the gross nominal interest rate paid on deposits to households.⁹

As described by equation (14), the maximum amount of lending financial intermediaries can obtain depends on the total net worth, \hat{N}_t and on the ratio of loan asset to equity capital, \hat{lev}_t . The leverage is endogenously determined by equation (15) and depends on the gain of increasing one unit of net worth, \hat{D}_t , on the gain of expanding assets, \hat{V}_t , and on the fraction λ that bankers could divert from the project and transfer it back to their household. The gain from having net worth, equation (16), hinges on the stochastic discount factor, Λ_t , associated with the household problem, the probability of bankers' surviving in the next period, θ , and on the gross growth rate of net worth, $\hat{Z}_{t,t+1}$, the law of motion of which is given by equation (17). The gain of expanding assets, equation (18), is mainly affected by the gross growth rate in assets, $\hat{X}_{t,t+1}$, which evolves as in equation (19).

Total net worth is given by the sum of net worth of existing bankers, \hat{N}_t^e and of new bankers, \hat{N}_t^n , in equation (20). The net worth of existing bankers equals earnings on assets held in the previous period and the growth of the net worth, as specified by equation (21), while the net worth of new banks, equation (22), takes into account the "start-up" funds from the households they belong to, equal to the fraction χ of total assets.¹⁰

(iv) Monetary Authority and Equilibrium

$$\hat{R}_{t}^{n} = \rho_{i}\hat{R}_{t-1}^{n} + (1 - \rho_{i}) \left[\rho_{\pi}\hat{\Pi}_{t} + \rho_{y} \left(\hat{Y}_{t} - \hat{Y}_{t}^{p} \right) \right] + \rho_{\Delta y} \left[\hat{Y}_{t} - \hat{Y}_{t}^{p} - \left(\hat{Y}_{t-1} - \hat{Y}_{t-1}^{p} \right) \right] + \varepsilon_{t}^{r}$$
(23)

$$\hat{Y}_t = \frac{C}{Y}\hat{C}_t + \frac{I}{Y}\hat{I}_t + \frac{G}{Y}\varepsilon^g_t + Z^k\frac{K}{Y}\hat{U}_t$$
(24)

The monetary authority follows a Taylor rule, equation (23), where ρ_i , ρ_{π} , ρ_y and $\rho_{\Delta y}$ are policy parameters referring to interest-rate smoothing, and the responsiveness of the nominal interest rate to inflation deviations, to the output gap and to changes in the output gap, respectively. The term \hat{Y}_t^p represents the level of output that would prevail under flexible prices and wages without the two mark-up shocks, while ϵ_t^r is a monetary policy shock. Finally, the resource constraint, equation (24), completes the model, with ϵ_t^g being a government spending shock.

⁹In this model at the beginning of each period the banker can choose to divert the fraction λ of available funds from the project and transfer them back to the household. Depositors can force the intermediary into bankruptcy and recover the remaining fraction $1 - \lambda$ of total assets. However, costly enforcement implies that it is too costly for the depositors to recover the diverted fraction of funds by the banker.

¹⁰While parameter χ affects the steady-state of the model, it is not featured in the log-linearized version of the definition new banks' net worth.

Each of the nine exogenous shocks follow an AR(1) process,¹¹ except the price and wage mark up shocks which follow ARMA(1,1) processes. The inclusion of the MA term in these two exogenous disturbances allows capturing high-frequency fluctuations in inflation. This specification is common in the DSGE literature (e.g., Smets and Wouters, 2007; Justiniano et al., 2010, among many others).

3 Estimation

This section reports the results of the Bayesian estimation. Section 3.1 discusses the data and the estimation strategy while Section 3.2 discusses the calibration and presents parameter estimates.

3.1 Data and Measurement Equations

The model is estimated with quarterly data for the period 2000Q1-2022Q1, using as observables: (i) real GDP, (ii) real investment, (iii) real private consumption, (iv) the capital utilization rate, (v) hours worked, (vi) GDP deflator inflation, (vii) real wage, (viii) the shadow nominal interest rate, (ix) the interest rate spread and (x) the net worth of financial intermediaries. The starting date is dictated by the availability of data on EA net worth.

Most of EA data come from the Area Wide Model database (see Fagan et al., 2005, for details), complemented by capital utilization in manufacturing from the ALFRED database of the Federal Reserve Bank of St. Louis, the net worth extracted from Euro Area Statistics, the interest rate spread computed by Gilchrist and Mojon (2018), and hours worked, which combine observations computed by Ohanian and Raffo (2012) until 2016Q4 and from the Eurostat dataset afterwards.

Most of the US data are extracted from the ALFRED database, complemented with the series of the credit spread computed by Favara et al. (2016). All series are seasonally adjusted by their sources.

GDP, consumption, investment and wages are logged and, together with capital utilization, are expressed in first differences. Hours worked are logged and demeaned, while net worth is expressed as a ratio of GDP. The inflation rate is measured as the quarterly log-difference of the GDP deflator. Consistently, nominal interest rate and the spread are expressed in quarterly terms. More granular details on data sources and transformations are reported in Appendix A.

¹¹The literature adopts a variety of approach in modelling monetary policy shocks (see, e.g. Taylor and Wieland, 2012). We follow Carrillo et al. (2018), who find that, when the framework contains enough information, a policy rule with interest rate inertia and serially correlated shocks satisfactorily matches the data.

The following set of measurement equations show the link between the observables in the dataset and the endogenous variables of the DSGE model:

$$\begin{bmatrix} \Delta Y_{t}^{o} \\ \Delta C_{t}^{o} \\ \Delta I_{t}^{o} \\ \Delta I_{t}^{o} \\ \Delta W_{t}^{o} \\ \Delta W_{t}^{o} \\ \frac{\Delta U_{t}^{o} \\ L_{t}^{o} \\ \frac{N^{o}}{\overline{Y}} \\ \frac{N^{o}}{\overline{Y}} \\ \frac{N^{o}}{\overline{r}_{t}^{n}} \\ \frac{R^{n,o}}{r_{t}^{n}} \\ EP_{t}^{o} \end{bmatrix} = \begin{bmatrix} \gamma \\ \gamma \\ \gamma \\ \overline{\gamma} \\ \overline{\eta} \\ \frac{\overline{\eta}}{\overline{\eta}} \\ \frac{\overline{\eta}}{\overline{\eta}} \\ \frac{\overline{\eta}}{\overline{y}} \\ \frac{\overline{\eta}}{\overline{\eta}} \\ \frac{\overline{\eta}}{\overline{r}} \\ \frac{\overline{r}^{n}}{\overline{r}} \\ \frac{\overline{r}^{n}}{\overline{EP}} \end{bmatrix} + \begin{bmatrix} Y_{t} - Y_{t-1} \\ \hat{L}_{t} \\ 0 \\ \frac{\overline{\eta}}{\overline{\eta}} \\ \frac{\overline{\eta}}{\overline{\eta}} \\ \frac{\overline{\eta}}{\overline{\eta}} \\ \frac{\overline{\eta}}{\overline{R}} \\ \frac{\overline{R}^{n}}{\overline{EP}} \end{bmatrix} + \begin{bmatrix} V_{t} - Y_{t-1} \\ \hat{U}_{t} - \hat{U}_{t-1} \\ \hat{U}_{t} - \hat{U}_{t-1} \\ \frac{1}{\hat{L}_{t}} \\ \frac{\overline{\eta}}{\overline{\eta}} \\ \frac{\overline{\eta}}{\overline{\eta}} \\ \frac{1}{\overline{\eta}} \\ \frac{1}{\overline{R}} \\ \frac{1}{\overline{R}} \\ \frac{1}{\overline{R}} \\ \frac{1}{\overline{R}} \\ \frac{1}{\overline{EP}} \end{bmatrix} + \begin{bmatrix} Y_{t} - Y_{t-1} \\ \hat{U}_{t} - \hat{U}_{t-1} \\ \frac{1}{\hat{L}_{t}} \\ \frac{1}{\overline{\Omega}} \\ \frac{1}{\overline{\Omega}} \\ \frac{1}{\overline{\Omega}} \\ 0 \\ 0 \\ 0 \\ 0 \\ \varepsilon \\ \frac{EP}{EP} \end{bmatrix}$$
(25)

where γ is the common quarterly trend growth rate of GDP, consumption, investment and wages; \bar{u} , \bar{L} and $\frac{\bar{N}}{Y}$ are the average change in capital utilization, average hours worked and the average ratio of net worth of financial intermediaries to GDP, respectively; $\bar{\pi}$, \bar{r}^n and \bar{EP} are the steady-state quarterly inflation rate, the nominal interest rate and spread, respectively. The term ε_{EP}^{me} represents a measurement error in the equation for the spread, which accounts for a possible mismatch between the financial variables in the model and those in the data, analogously to what Castelnuovo and Nisticó (2010) do for stock prices, among others.

The choice of including the utilization rate shock and the capital utilization in manufacturing as an observable variable stems from the inability of the canonical model to endogenously generate the large movements of the series itself observed in the data. Figure 1 shows the (demeaned) series of EA and US capital utilization in the data, its one-step-ahead forecast produced by the baseline model, and a counterfactual model estimated without capital utilization shocks. The baseline model helps forecast the utilization rate in a much more compelling way than the counterfactual model. While the counterfactual model is still able to generate pro-cyclical series, it displays a considerably smaller volatility, and is unable to match the large contractions observed during the crisis periods.

3.2 Calibration and Estimates

The parameters that cannot be identified in the data and/or are related to steady-state values of endogenous variables are calibrated. Given that quarterly data are employed for the estimation, the frequency implied by the calibrated parameters is also quarterly. Table 1 reports the parameters common to both countries and those taking country-specific values.

The common parameters are assigned very standard values in line with the DSGE

Figure 1: Capital Utilization in the Data (Demeaned), in the Baseline Model (One-Step-Ahead Forecast) and in a Counterfactual Model Without Capital Utilization Shocks (One-Step-Ahead Forecast).



(a) Euro Area

literature. The capital depreciation rate, δ , is set to 0.025, corresponding to an annual depreciation rate of 10 percent. The discount factor, β , is set to 0.99 and capital share of income, α , is equal to 0.33. The elasticities of substitution in goods and labor markets,

Parameters	Value		Steady-state target/reference	
Common	Euro area United States			
Capital deprec. rate, δ	0.025		10% depreciation rate p.a. (standard)	
Discount factor, β	0.99		4%risk-free real rate p.a. (standard)	
Cap. share of income, α	0.33		labour share $2/3$ of income (standard)	
Elast. of subst. goods, ε	of subst. goods, ε 6		mark-up of 20% (Christiano et al., 2014)	
Elast. of subst. labor, ε_w	6		mark-up of 20% (Christiano et al., 2014)	
Surv. rate of bankers, θ	0.969		avg life of 8 yrs (Gelain and Ilbas, 2017)	
Country-specific				
Gov. spend. to GDP, g_y	0.193	0.205	OECD and NIPA tables data	
Frac. assets new banks, χ	0.003	0.003	leverage of 3.75 (Gelain and Ilbas, 2017)	
Frac. of divert. assets, λ	0.683 0.393		spread 107; 203 a.bps. (Gilchrist and Mo- jon, 2018; Gilchrist and Zakrajšek, 2012)	

 Table 1: Calibrated Parameters

 ε and ε_w are equal to 6 in order to target a gross steady-state mark-up of 1.20, as in Christiano et al. (2014), among many others. The survival rate of financial intermediaries, θ , is set equal to 0.969 to target an average life of 8 years, as common in studies employing the Gertler-Karadi setting for the financial sector (as, e.g., in Gelain and Ilbas, 2017).¹²

As regards the country-specific parameters, the government spending to GDP ratio, g_y , is set equal to 20.7 percent for the EA and to 18.8 percent for the US economy, in line with OECD and NIPA tables data, respectively. The parameter representing the fraction of assets given to the new bankers, χ , is set equal to 0.003 for the EA and to 0.001 for the US economy, while the fraction of divertible assets, λ , is set equal to 0.393 for the EA and to 0.663 for the US economy. The combination of these two parameters and the rest of the calibration yield a leverage of 3.75 in both countries (as, e.g., in Gelain and Ilbas, 2017) and a steady-state credit spread of 107 and 203 annual basis points for the EA and the US economy, respectively, in line with the data (Gilchrist and Mojon, 2018; Gilchrist and Zakrajšek, 2012).

Table 2 shows the assumptions for the prior distributions, which are the same for both countries. This approach is line with other studies in the literature (e.g., Lubik and Schorfheide, 2005; Smets and Wouters, 2005; Giovannini et al., 2019, among others) and is meant to let the data determine cross-country differences in the estimated parameters. The table reports the prior mean and standard deviation of almost all parameters. The only exception is the standard error of innovations, for which the degrees of freedom of

¹²The parameter is not directly observable in the data. However, indicators related to bank failures suggest that it has not changed significantly during the C-19C, both in the EA and the US economy (see, e.g., the number of failed U.S. institutions reported by the Federal Deposit Insurance Corporation and the probability of simultaneous default of two or more large banks available in the ECB Statistical Data Warehouse). Moreover, this paper assumes the same magnitude for parameter θ in the two economies, given that contributions employing EA and US data adopted similar values (Gertler and Karadi, 2011; Hirakata et al., 2013; Lim and McNelis, 2016; Sahuc, 2016; Coenen et al., 2018; Quint and Tristani, 2018).

Prior di			or distribut	tion	Posterior mean	
Parameters	Distr	Mean	Std./df	Euro area	United States	
Structural parameters				,		
Habit parameter	h	Beta	0.7	0.1	0.13 [0.10;0.16]	0.12 [0.10; 0.13]
Inv. of Frisch elasticity	ϕ	Gamma	0.33	0.25	0.22 [0.20;0.24]	0.41 [0.17;0.63]
Calvo prices	σ_p	Beta	0.5	0.05	0.75 [0.72;0.79]	0.71 [0.66;0.76]
Calvo wages	σ_w	Beta	0.5	0.05	0.31 [0.28;0.34]	0.78 [0.65;0.91]
Price indexation	σ_{pi}	Beta	0.5	0.15	0.16 [0.05;0.26]	0.57 [0.50; 0.64]
Wage indexation	σ_{wi}	Beta	0.5	0.15	0.47 [0.27;0.67]	0.67 [0.46;0.88]
Inv. adj. costs	ξ	Normal	4	0.5	3.21 [3.00;3.44]	3.54 [3.07;4.00]
Elasticity of capital util	ζ	Beta	0.25	0.15	0.62 [0.53;0.71]	0.64 [0.56;0.71]
Inflation - Taylor rule	ρ_{π}	Normal	1.7	0.15	2.05 [1.86;2.25]	1.78 [1.58;1.97]
Output - Taylor rule	ρ_y	Gamma	0.125	0.05	0.02 [0.01;0.03]	0.11 [0.07;0.14]
Taylor rule changes in y	ρ_{Δ_y}	Normal	0.0625	0.05	0.11 [0.07;0.14]	0.08 [0.05;0.11]
Taylor rule smoothing	ρ_i	Beta	0.75	0.1	0.67 [0.60;0.74]	0.76 [0.71;0.80]
Constants	Γl			0.2	[]	
Trend	$ar{\gamma}$	Normal	0.4	0.2	0.02 [0.00;0.04]	0.29 [0.25;0.32]
Inflation	π	Gamma	0.5	0.1	0.58 [0.48;0.69]	0.66 [0.47;0.83]
Interest rate	\bar{R}	Normal	0.8	0.2	0.31 [0.12;0.49]	0.33 [0.13; 0.52]
Hours	$\bar{\ell}$	Normal	0.0	2.0	-0.12 [-2.21;1.88]	3.45 [2.04;5.00]
Utilization rate	\overline{U}	Normal	0.0	0.2	-0.27 [-0.30;-0.24]	0.01 [-0.02;0.04]
Spread	\overline{S}	Gamma	0.5	0.1	0.37 [0.28;0.46]	0.44 [0.36;0.52]
Net worth over GDP	\tilde{NY}	Normal	0.05	0.2	0.08 [-0.25;0.40]	0.06 [-0.26;0.38]
Exogenous processes	1.1	1.011101	0.000	0	0.000 [0.20,0.10]	0.000 [0.20,0.000]
Technology	$ ho_a$	Beta	0.5	0.2	0.93 [0.88;0.97]	0.84 [0.78;0.91]
10011101085	σ_a	IG	0.0	2	0.85 [0.74;0.96]	0.75 [0.65;0.85]
Price mark-up	ρ_p	Beta	0.5	0.2	0.99 [0.97;1.00]	0.91 [0.87;0.95]
i nee man ap	$\sigma_p^{ ho p}$	IG	0.0	2	0.28 [0.23;0.34]	0.24 [0.19;0.29]
	μ_p	Beta	0.5	0.2	$0.51 \ [0.39; 0.62]$	0.69 [0.58;0.80]
Labor	ρ_w	Beta	0.5	0.2	0.97 [0.95;0.99]	0.98 [0.96;0.99]
	$\sigma_w^{ how}$	IG	0.0	2	0.30 [0.24;0.35]	$0.72 \ [0.59; 0.85]$
	μ_w	Beta	$0.1 \\ 0.5$	0.2	0.18 [0.05;0.30]	$0.75 \ [0.65; 0.84]$
Capital utilization	$ ho_{u}$	Beta	0.5	0.2	0.92 [0.89; 0.95]	0.84 [0.77;0.91]
Capital atilization	$\sigma_u^{ ho u}$	IG	0.0	2	2.82 [2.30; 3.32]	1.83 [1.54; 2.12]
Capital quality		Beta	$0.1 \\ 0.5$	0.2	0.96 [0.94;0.98]	$0.74 \ [0.68; 0.79]$
Capital quality	$ ho_{kq}$	IG	0.5 0.1	2	$0.12 \ [0.09; 0.15]$	0.36 [0.28; 0.45]
Inv. specific	σ_{kq}	Beta	0.1	0.2	0.12 [0.09; 0.13] 0.19 [0.09; 0.28]	$0.39 \ [0.28; 0.49]$
my, specific	$ ho_x$ σ	IG	$0.3 \\ 0.1$	$\frac{0.2}{2}$	$7.50 \ [6.21; 8.78]$	4.71 [3.72; 5.67]
Preference	σ_x	Beta	$0.1 \\ 0.5$	$\frac{2}{0.2}$	0.77 [0.69; 0.86]	4.71 [5.72, 5.07] 0.99 [0.99; 0.99]
	$ ho_b \ \sigma_b$	IG	$0.3 \\ 0.1$	$\frac{0.2}{2}$	$1.12 \ [0.90; 1.32]$	3.28 [2.74; 3.85]
Monetary policy		Beta	$0.1 \\ 0.5$	0.2	$0.35 \ [0.25; 0.44]$	$0.45 \ [0.38; 0.52]$
monetary poncy	$ ho_m$	IG	$0.3 \\ 0.1$	$\frac{0.2}{2}$	0.35 [0.23; 0.44] 0.27 [0.22; 0.32]	$0.45 \ [0.38; 0.52]$ $0.19 \ [0.16; 0.22]$
Government spending	σ_m	IG Beta	$0.1 \\ 0.5$	0.2	0.27 [0.22; 0.32] 0.63 [0.48; 0.78]	0.19 [0.10; 0.22] 0.91 [0.87; 0.96]
Government spending	$ ho_g$	IG	$0.3 \\ 0.1$	$\frac{0.2}{2}$	1.94 [1.70;2.18]	2.94 [2.58; 3.30]
Std- Measurement error	σ_g	IG IG	$0.1 \\ 0.1$	$\frac{2}{2}$	$0.15 \ [0.13;0.17]$	2.94 [2.58; 5.50] 0.15 [0.13; 0.17]
Su- measurement error		10	0.1	4	0.10 [0.13,0.17]	0.10 [0.10,0.17]

Table 2: Prior and Posterior Distributions of Estimated Structural Parameters, Constants and Exogenous Processes (95 percent credible intervals in square brackets)

the inverse Gamma distribution are reported. In line with many studies conducting a Bayesian estimation of DSGE models (e.g. Smets and Wouters, 2003, 2005, 2007, among many others), choosing 2 degrees of freedom guarantees a large domain for the prior of

the volatility of all shocks. The functional form and the prior mean of the distribution of all parameters and constants largely correspond to those available in the literature (see, e.g. Smets and Wouters, 2003, 2007; Kollmann et al., 2015; Villa, 2016; Gelain and Ilbas, 2017; Albonico et al., 2019b).

The mean of the estimated parameters for each model is computed with two chains of the Metropolis–Hastings algorithm, each with a sample of 500,000 draws. Table 2 reports the posterior mean with 95 percent probability intervals in parentheses for the EA and the US economy. Posterior estimates are broadly in line with previous studies. Due to the presence of financial frictions, the estimated value of the habit parameter in consumption is lower than in other papers (Smets and Wouters, 2007). This finding is in line with several contributions in the literature (e.g., De Graeve, 2008; Smets and Villa, 2016), since financial frictions generate enough endogenous propagation to account for the persistence of the consumption process. The response of the nominal interest rate to inflation is higher in the EA than in the US economy, while the policy rate is more aggressive to output in the US economy compared to the EA. The estimates of the exogenous processes reveal that, in general, shocks are more volatile in the US economy compared to the EA, except for capital utilization, investment-specific technology and monetary policy shocks.¹³ The persistence of the monetary policy shock is relative low, similarly to Smets and Wouters (2007). As far as the capital utilization shocks are concerned, these are rather persistent, capturing empirical features in the data. The posterior estimates of constants in the measurement equations capture the sample averages of the observable times series used in the estimation.¹⁴

For some parameters, such as the habit parameter and the responsiveness of the nominal interest rate to the output gap, the estimates are close to the extremes of the prior distributions. We investigate whether a different prior distribution can affect the estimation results (see, for example, Meenagh et al., 2022) in Appendix C.2, which reports posterior estimates, as well as all the results on the role of shocks on economic activity and inflation, under an alternative specification of the prior distributions of relevant parameters.

4 Results

This section presents the results of the paper. Subsection 4.1 discusses the dynamic properties of the model via an analysis of the estimated impulse response functions. Subsection 4.2 investigates the sources of business cycles fluctuations during the GFC

 $^{^{13}}$ As discussed in Smets and Wouters (2005), if the posterior estimate of a parameter in one model falls in the estimated confidence band for the same parameter of the other model, the two estimated parameters can be considered similar.

¹⁴These may vary substantially across the two regions because of the underlying differences in the data, as in the case of trend growth, hours worked and the utilization rate.

and the C-19C with historical shock decompositions. Finally, Subsection 4.3 focuses on the contribution of monetary policy shocks.

4.1 Dynamic Properties of the Estimated Model

As mentioned in Subsection 3.1, the estimated model features nine structural shocks, most of which are common in the DSGE literature (e.g. Smets and Wouters, 2007). One notable exception is the capital utilization shock. While endogenous capital utilization is a rather standard element in this class of models (e.g. Smets and Wouters, 2007; Albonico et al., 2014), exogenous disturbances to this variable have so far been neglected. The C-19C triggered by the COVID-19 pandemic involved several necessary containment measures impairing the use of existing capital goods, which can be captured by an exogenous disturbance to capital utilization.

Another important implication of the C-19C was the inability of a large fraction of workers to reach their workplace or to provide their usual amount of labor services remotely. This aspect is captured by a shock to labor supply introduced as a standard wage mark-up shock. This shock is observationally equivalent to the labor disutility disturbance in terms of dynamics because both shocks enter households' intratemporal optimality condition (e.g. Smets and Wouters, 2007).¹⁵

The inclusion of financial frictions and of a financial shock, such as the capital quality shock as in Gertler and Karadi (2011), is warranted primarily because the estimation sample includes the GFC, but also to capture disturbances to the financial sector in the rest of the sample.

Both the capital quality shock and the capital utilization shock affect the amount of capital that effectively enters the production function. It is, therefore, important that the responses of the model's endogenous variables to these two shocks display sufficiently different dynamics for them to be correctly identified. Figure 2 reports the posterior median impulse response functions (IRFs), for the EA and the US economy, of important macroeconomic variables to a capital quality and a capital utilization shock of sizes equal to their respective estimated standard deviations. The sign of the shocks is such that they trigger a decline in output. Both shocks cause a fall in hours worked and an increase in the credit spread. The crucial difference between the two lies in the response of inflation, which declines in the case of a capital quality shock and rises in that of the capital utilization shock. In other words, while the capital quality shock behaves as a demand shock,¹⁶ the capital utilization shock acts as a supply shock by affecting the production

¹⁵Some contributions have disentangled the two shocks by imposing different spectral profiles (Justiniano et al., 2013) or by introducing unemployment (Galí et al., 2012; Foroni et al., 2018).

¹⁶The mechanism is the following: the capital quality shock directly translates into a shock to banks' balance sheets because of the identity between capital and assets. Because of the presence of moral hazard, depositors require banks not to be overleveraged. Hence they are forced to reduce lending. The reduction in the lending volume makes the incentive constraint tighter, leading to a higher spread.



Figure 2: Posterior Median Impulse Response Functions to One-Standard-Deviation Capital Quality and Capital Utilization Shocks

function. The monetary policy rate closely tracks the path of inflation due to the high weight inflation takes in the estimated Taylor rules. The typical responses of output to both shocks are more pronounced for the EA than they are for the US economy, largely due to their higher estimated volatility and/or persistence.

As regards the other more standard structural shocks, their IRFs are reported in Appendix B (Figure B.1 and B.2). The dynamics of the IRFs of output, inflation and the monetary policy rate have the expected sign and shape, as well as relatively narrow

Nonfinancial firms observe a rise in borrowing costs and consequently reduce their demand for capital, leading to a fall in investment.

confidence bands (at a 95-percent confidence level). While the shocks to investment, preferences, monetary policy and government spending behave as demand shocks, the responses to shocks to TFP, price mark-up and labor supply display dynamics in line with a supply shock.

4.1.1 Investigating Heterogeneity

While the core of this paper investigates aggregate results for the EA and the US economy as a whole, this subsection examines potential cross-country or cross-states heterogeneity. For example, some empirical studies for the EA (Cortes et al., 2022; Spiegel, 2022) find heterogeneous impacts of policy shocks on economic and financial outcomes. As far as monetary policy shocks are concerned, Corsetti et al. (2022) emphasize the ECB's "one money, many market" feature of monetary policy decisions. Also for the US economy, there is evidence of heterogeneous impacts of monetary policy (Beraja et al., 2019; Amir-Ahmadi et al., 2020), while Brinca et al. (2021) emphasize differences at the sectoral level. Unlike currency-union models (e.g. Batini et al., 2021), the model employed in this paper does not have a multi-country or a multi-sectoral structure. However, alternative parametrizations may help shed light on the role of regional or sectoral heterogeneity in the transmission of shocks. Given the novelty of capital utilization shocks and the abovementioned debate on monetary policy shocks, this subsection focuses precisely on these disturbances. The choice of parameters is motivated by characteristics that may affect more directly the transmission of these shocks. In particular, given the estimated parameters, we investigate whether IRFs are stable when changing a relevant parameter at a time.

Gautier et al. (2022), using micro-data, find that while differences in price rigidity for most EA countries are small, there are also cases of significantly lower price rigidity. The average quarterly frequency of price changes found by these authors is 29.5 percent, which translates into a Calvo parameter of $\sigma_p = 0.71$, equivalent to changing prices almost every 3.5 quarters on average. This estimate, although obtained using a very different empirical approach, is close to the DSGE-based point estimate presented in this paper ($\sigma_p = 0.75$, equivalent to changing prices every 4 quarters on average). In the countryspecific estimates by Gautier et al. (2022), the minimum probability of changing prices corresponds to resetting them every two quarters ($\sigma_p = 0.5$). Therefore, Figure 3(a) shows a comparison of impulse responses to a capital utilization shock and a monetary policy shock between the baseline case (in line with aggregate EA estimates presented in Table 2) and an alternative specification with $\sigma_p = 0.5$.¹⁷

As regards the investment adjustment cost parameter, ξ , the DSGE literature using

¹⁷Alternatively, a parameter potentially capturing regional heterogeneity in the EA could be wage indexation to past inflation, which differs across states (ECB, 2008). This parameter affects the dynamics of the monetary policy shock, having virtually no role in the transmission of the capital utilization shock.

Figure 3: Impulse Responses to Capital Utilization and Monetary Policy Shocks Under Alternative Parametrizations



Baseline – – Lower Investment adj. costs Guasi-flexible prices

estimated or calibrated models employing EA data (at the level of monetary union or single countries) generally report values in a neighborhood of our estimate of 3.21. For example, Forni et al. (2010) set $\xi = 3.5$ for Belgium, Germany and the rest of the EA, while Drygalla et al. (2020) estimate $\xi = 3.91$ using German data. There are, however, some studies that estimate or calibrate this parameter at considerably different values, indicating that there may be some heterogeneity owing to the use of a different country coverage and/or a different estimation/calibration period. For instance, Breuss and Rabitsch (2009) estimate $\xi = 0.86$ and $\xi = 1.76$ for Austria and the EA, respectively; Poutineau and Vermandel (2015) estimate $\xi = 0.63$ and $\xi = 1.87$ for core and peripheral EA, respectively; and Gerali et al. (2018) calibrate $\xi = 1$ both for Italy and the EA. Therefore, Figure 3(a) also shows a comparison of impulse responses between the baseline case and an alternative specification with $\xi = 0.63$, which is the value (among those cited) that differs the most from the EA estimate of 3.21 presented in Table 2.

Results confirm the abovementioned studies arguing that intra-EA heterogeneity may matter for the transmission of some shocks. For example, price rigidity has a considerable impact in the case of capital utilization and monetary policy shocks. In countries with more flexible prices, the real effects of a monetary policy shocks are dampened, in line with a large literature on monetary economics (e.g. Woodford, 2003). Conversely, output contracts to a larger extent in response to a negative capital utilization shock if prices are more flexible. In countries with lower investment adjustment costs, the effects of monetary policy shocks may be amplified due to the higher investment responsiveness. This feature does not have a big impact in the case of capital utilization shocks.

While for the EA there are several contributions on the heterogeneity of price stickiness across member countries, for the US economy estimates are available at the aggregate national level. However, regional differences, such as varying inflation rates, industry compositions, and labor market conditions, could impact the level of price stickiness at the state level. For instance, Nakamura and Steinsson (2008) investigate the heterogeneity at the sector level and find that the (approximated) quarterly frequency of price changes ranges from 10 percent in the case of apparel to 99 percent in the case of vehicle fuel, equivalent to Calvo parameters of $\sigma_p = 0.90$ and $\sigma_p = 0.01$, respectively. In other words, across sectors, prices range from being as sticky as not changing for 10 quarters (on average) to almost flexible (within the quarter). As an illustration, Figure 3(b) shows a comparison of impulse responses between the baseline case and an alternative specification with $\sigma_p = 0.01$, which is the value (among those cited) that differs the most from the US estimate of 0.71 presented in Table 2.

Also for the investment adjustment cost parameter, there is no study employing a DSGE framework investigating cross-regional differences. However, there is some empirical evidence showing that these costs vary across sectors (Groth and Khan, 2010) and across types of agents (Iacoviello, 2015). Therefore, it is plausible that diverse sectoral compositions and/or agents' distributions across states also imply different levels of investment adjustment costs. Cantelmo and Melina (2018) build a two sector DSGE model with durable and non-durable good, and patient and impatient agents, where the baseline estimates of the investment adjustment cost parameter range between $\xi = 1.77$ and $\xi = 3.79$. As an example, Figure 3(b) shows impulse responses obtained with an alternative specification of ξ equal to 1.77, which is the value that differs the most from the US estimate of $\xi = 3.54$ presented in Table 2.

Results found with different specifications for the US economy are qualitatively similar to those obtained using EA alternative specifications. The model predicts that, in states with a sector composition biased toward categories of goods and services with more flexible prices, the effects of capital utilization shocks are expected to be stronger than those at the average national level. For monetary policy shocks, the reverse is true, given the well-known feature of monetary policy affecting more the demand of goods and services with stickier prices. In line with EA findings, in states with lower investment adjustment costs, the effects of monetary policy shocks may be enhanced, while this feature is less relevant for capital utilization shocks.

4.2 The Role of Shocks on Economic Activity and Inflation

This subsection focuses on the historical contribution of structural shocks in explaining quarterly output growth and inflation fluctuations in the period of the GFC and of the C-19C in order to provide a comparison between the two recessions across the two economies. In Figures 4 and 5, each color represents the contribution of an individual shock. Two exceptions, implemented to simplify the charts, are the green bars, which combine the TFP and the price mark-up shocks, and the light blue bars merging (monetary and fiscal) policy shocks.

4.2.1 Output

The historical decomposition of quarterly GDP growth is shown in Figure 4. At the trough of the GFC, in the EA the capital utilization shock is the largest contributor to output growth, while policy shocks provided the second largest contribution, as reported in Figure 4(a). Financial shocks also played a non-negligible role, persistently weighing on GDP growth for about one year after the trough. Over the GFC, capital utilization shocks confirmed their dominant role together with the other two supply shocks and the financial shocks. These results broadly agree with those by Kollmann et al. (2016) and Cardani et al. (2022), who estimated much richer DSGE models for the EA, finding a non-negligible role also for trade shocks, absent in the closed-economy model employed in this paper. During the GFC, foreign demand shocks played a relevant role particularly for some countries, such as France (Albonico et al., 2019a).

At the trough of the C-19C (Figure 4(b)), the capital utilization shock and the other supply shocks account for half of the output contraction, in line with Kollmann (2021) who

Figure 4: Historical Shock Decomposition of Quarter-on-Quarter GDP Growth (Demeaned)



(a) Euro Area GFC

(b) Euro Area C-19C

finds that an aggregate supply contraction is identified as the dominant force driving the sharp fall of EA GDP in 2020. Investment-specific technology shocks played a noteworthy role while labor and preference shocks explain only a minor fraction of the total variation in output growth.

In the US, at the trough of the GFC, the capital utilization shock was one of the main driver of the output contraction together with the labor (or wage mark-up) shock, as shown in Figure 4(c). The latter was much more important in the US economy than in the EA, confirming the results by Smets and Wouters (2005) over an earlier sample (1975Q1-2002Q2). The negative prolonged role of the labor supply shocks is consistent with the fact that the recovery from this crisis has been defined "jobless" (see Cantore et al., 2014, among others). The larger role of labor shocks can be explained by labor market variables in the US economy being more reactive to business cycle fluctuations, given the looser employment protection than in the EA (e.g. Nunziata, 2003; Gnocchi et al., 2015). The rate of labor reallocation into and out of unemployment observed in the US is indeed much higher than that observed in European countries (e.g. Elsby

et al., 2009) and movements in the labor supply are likely to affect output fluctuations to a large extent. Policy shocks contributed notably to support GDP growth at the trough of the GFC. Surprisingly, financial shocks played a limited role in accounting for GDP fluctuations, while other shocks had a more prominent role. There are two possible explanations. First, investment-specific technology shocks dominate in 2009Q1 in correspondence with the massive drop observed in investment. Second, shocks to capital utilization are found to be particularly relevant in explaining the large fall in the utilization rates observed during the GFC, as shown in Figure 1b.

At the trough of the C-19C (Figure 4(d)), labor and preference shocks explained half of the output fluctuations. In line with Christiano et al. (2011), preference shocks capture exogenous disturbances affecting desired savings. Saving rates have indeed reached unprecedented levels during the pandemic, and this explains its important role on business cycle fluctuations. Other supply shocks drove output up, and capital utilization shocks played a relevant role similarly to the GFC.¹⁸ In line with purely empirical papers (see, e.g., Christiano et al., 2005; Ramey and Zubairy, 2018, among many others), the estimated model implies lagged effects of monetary and fiscal policy. This explains why a positive GDP effect of policy shocks is detectable only with lags.

4.2.2 Inflation

Figure 5 shows the shock decomposition for inflation. Over the GFC, EA financial shocks played an important role in driving inflation fluctuations (Figure 5(a)). Interestingly, financial shocks contributed positively to inflation for the four quarters of 2008, then their contribution turned negative. There is a large discussion in the literature on whether financial shocks act as a demand or a supply shock (e.g. Gerali et al., 2010; Meh and Moran, 2010; Furlanetto et al., 2019, among others), as well as on the missing deflation puzzle in the GFC (see Kara and Pirzada, 2020; Harding et al., 2022, among others).¹⁹ In this model, similarly to Gerali et al. (2010), inflationary pressures may originate from the dynamics of bank lending standards, the easing of which in 2006 and 2007 could have had persistent effect on inflation due to the high estimated persistence of the financial shock. From 2009 onward, in combination with higher borrowing costs and a tightening of credit standards, financial shocks contributed negatively on inflation. Supply shocks

¹⁸The positive contribution of other supply shocks at the trough of the C-19C, followed by smaller contributions in subsequent quarters, is broadly in line with some empirical studies on the US economy (e.g., Fernald and Li, 2022) arguing for accelerated productivity growth at the beginning of the pandemic followed by more modest increases afterwards. One interpretation is the fast adoption of digital technology associated with teleworking. This paper follows most of the DSGE literature in employing stationary data for the estimation, which may downplay the contribution of TFP shocks, found to be sizable by Le et al. (2021) who use non-stationary data in the estimation.

¹⁹In a DSGE model estimated for the EA, Gerali et al. (2010) also find that the role of financial shocks turned negative from positive in accounting for movements in inflation during the financial crisis. In a BVAR estimated on US data, Abbate et al. (2016) find that contractionary financial shocks temporarily increase inflation, thereby helping explain the missing disinflation during the GFC.

Figure 5: Historical Shock Decomposition of Quarter-on-Quarter Inflation Rate (Demeaned)



(a) Euro Area GFC

(b) Euro Area C-19C

contributed mainly negatively to inflation. In particular, the positive contribution of the TFP shock was more than offset by the negative contribution of the price mark-up shock. The signs of these supply shocks are in line with the study by Kollmann et al. (2016). Both monetary and fiscal shocks contributed negatively to inflation fluctuations during the period 2008Q1-2011Q4. Their contribution turned large and positive around 2014 when the ECB embraced strongly accommodative policy actions.

During the C-19C reported in Figure 5(b), policy shocks contributed positively to EA inflation fluctuations, while financial shocks pushed inflation down, in line with the results by Cardani et al. (2022).²⁰ The positive effect of policy shocks is dominated by the contribution of monetary policy (displayed in Figure 7(a)). The expansionary phase of EA monetary policy started in 2015, it paused before the C-19C, and intensified in response to the pandemic. This pattern is mirrored in the contributions of policy shocks

 $^{^{20}}$ It should be noted that inflation is measured as quarter-on-quarter change in the GDP deflator. This can be particularly relevant when examining the role of monetary policy shocks in the EA during the C-19C, when the divergence between this measure of inflation and year-on-year HICP inflation gets very large.

to inflation with some lags, implied by their usual transmission mechanism. Supply shocks had a dominant and negative effect on inflation in 2020Q3, when the easing of restrictions over the summer can be interpreted as expansionary supply shocks reducing inflation and increasing GDP. The demand-side effects of easing restrictions initially had a much more muted impact on inflation, while they contributed positively toward the end of the estimation horizon.

Figure 5(c) includes the shock decomposition of inflation for the US over the GFC. Differently from the EA, there is an evident effect of contractionary demand shocks, mainly preference shocks (and financial shocks to a minor extent) which contributed negatively to inflation fluctuations; and contractionary supply shocks, mainly labor shocks, providing a positive contribution to inflation. Expansionary policy shocks, due to the timely intervention of policy-makers, pushed inflation up, though very little. Overall, demand shocks dominate over supply shocks in accounting for movements in inflation.

Lastly, during the C-19C reported in Figure 5(d), overall supply shocks dominated over demand shocks in explaining movements in inflation in line with the results by Del Negro et al. (2022). Policy shocks contributed positively to inflation from 2021Q1 onward.

4.3 The Contribution of Monetary Policy

This subsection focuses on the role of monetary policy shocks. Since the GFC, a major issue concerning both the EA and the US economy was the ZLB reached by the monetary policy rate, due to the repeated rate cuts. To overcome this constraint, the two central banks (as other monetary policy institutions around the world) pursued unconventional monetary policy measures, such as asset purchases and forward guidance. In the estimation, this enhanced policy intricacy is captured by using the shadow rate derived by Wu and Xia (2016, 2017, 2020).

As shown in Figure 6, the shadow rates declined during the GFC, capturing monetary policy easing that exerted a positive impact on GDP growth as early as 2008 in the US economy and as 2009 in the EA. Afterwards, the shadow rate path shows continued easing of US monetary policy until 2015, thanks to both the monetary policy rate being kept at the effective ZLB from 2009 to 2015 and various rounds of quantitative easing, mirrored in an expansion of the Fed's balance sheet. In the EA, monetary policy was tightened to an extent during the sovereign debt crisis that started in 2010. However, after the "Whatever it takes" speech of then ECB's President Mario Draghi, the monetary policy rate was brought first to the ZLB and then turned negative, before the Asset Purchase Program began in 2015.

It follows that, between 2007 and 2009—the period broadly coinciding with the GFC on average, monetary policy shocks had positive contributions to EA and US GDP growth

Figure 6: Contribution of Monetary Policy Shocks to Quarter-on-Quarter GDP Growth (Demeaned)



as shown in Table 3. The average annualized contribution of monetary policy shocks from the historical decomposition was 0.1 percentage points in the EA and 0.55 percentage points in the US economy. Slightly higher average contributions on annual GDP growth are found for the period between 2010 and 2014 (0.32 percentage points for the EA and 0.66 percentage points for the US economy). A key difference between the EA and the US is the period between 2015 and 2020. While the continued expansion of the ECB's balance sheet, as well as the additional non-standard measures, contributed positively to EA GDP growth (1 annual percentage point on average), in the US economy negative contributions dominate (-0.05 annual percentage points on average), due to repeated interest rate hikes between 2016 and 2019, mirrored in an increase in the shadow rate. Overall, these contributions are sizeable, especially if compared to the average annual growth rate of the two economies over the entire estimation period, which amount to

	Average contribution of monetary policy shocks to annual GDP growth				
	(Percentage Points)				
	Euro Area	United States			
2007Q1-2022Q1	0.61	0.31			
2007Q1-2009Q4	0.10	0.55			
2010Q1-2014Q4	0.32	0.66			
2015Q1-2022Q1	1.00	-0.05			
2009Q2-2011Q1	-0.67	0.43			
2020Q2-2022Q1	0.59	0.99			

Table 3: Contributions of Monetary Policy Shocks to Annual EA and US GDP Growth

Figure 7: Contribution of Monetary Policy Shocks to Quarter-on-Quarter Inflation (Demeaned)



1.27 and 2.02 percent for the EA and the US economy, respectively.

The greatest similarity that can be observed between the EA and the US monetary policy conduct is the swift and significant measures taken during the C-19C. The Fed brought the monetary policy rate again to the effective ZLB and expanded its balance sheet via a massive operation of bond purchases, and the ECB implemented a similar plan, the Pandemic Emergency Purchase Programme. These measures had the intended effects of boosting the recovery from the C-19C, contributing positively to GDP growth. In the two years from the trough of the C-19C (2020Q2-2022Q1) monetary policy shocks had an average contribution to annualized GDP growth of almost 0.6 and 1 percentage point in the EA and the US economy, respectively. In the EA context, this outcome stems from a more active policy-making approach compared to that adopted in the two years covering the Great Recession and beyond (2009Q2-2011Q1).²¹

The different timing in the monetary policy actions between the EA and the US economy is reflected also in their contributions to inflation (Figure 7). The EA shows mostly negative contributions of monetary policy shocks to inflation until the start of the ECB's Asset Purchase Program in 2015, and positive contributions until the end of the sample, including during the C-19C. In the US economy, monetary policy shocks had positive contributions to inflation during the GFC and for most of the period in which quantitative easing was in place. Contributions turned negative in 2015 and then again positive during the C-19C.

 $^{^{21}}$ The important role of the PEPP and other pandemic-related measures to support euro area growth in the period 2020-2023 is also found in ECB (2023).

5 Conclusions

The GFC and the C-19C represented not only two major economic global crises but also two important triggers of large and new policy actions. Understanding the drivers of the economic fallout around those episodes is essential to the analysis of the policy response also to future rare but large shocks. This paper focuses precisely on this issue by investigating the shocks that drove the output contraction and movements in inflation during the GFC and C-19C and by assessing to what extent monetary policy contributed to the recovery from these major crises.

The analysis is conducted with the Bayesian estimation of a DSGE closed-economy model augmented with financial frictions. A novel feature of the model is the addition of an exogenous disturbance to capital utilization, which proves important to replicate the capital utilization rate in crises periods. Given that the sample includes the ZLB period, the estimation is conducted employing a shadow monetary policy rate as an observable variable.

The main conclusions are as follows. Capital utilization shocks are found to be important contributors to the negative components of output growth during the GFC and the C-19C in both the EA and the US economy, while they played a limited role in accounting for inflation fluctuations. Moreover, the timely and strong intervention of the European Central Bank and the U.S. Federal Reserve is found to have had an important contribution to the recovery from the C-19C, having also a positive impact on inflation. It follows that the experience acquired by monetary policy makers during, and in the aftermath of, the GFC was arguably instrumental to the recovery from the C-19C.

Future research may investigate three main issues: (i) the role of open economy features in assessing the relative importance of various domestic and foreign shocks; (ii) the relevance of durable and non-durable goods, whose dynamics differed during the C-19C; and (iii) the large volatility of the COVID-19 shock, in the spirit of Lenza and Primiceri (2022). Enriching the model to include these features may shed light on additional mechanisms potentially relevant in the transmission of shocks in crisis times.

References

- Abbate, A., Eickmeier, S., and Prieto, E. (2016). Financial shocks and inflation dynamics. *Macroeconomic Dynamics*, pages 1–29.
- Albonico, A., Cales, L., Cardani, R., Croitorov, O., Ferroni, F., Giovannini, M., Hohberger, S., Pataracchia, B., Pericoli, F. M., Raciborski, R., Ratto, M., Roeger, W., and Vogel, L. (2019a). Comparing post-crisis dynamics across euro area countries with the global multi-country model. *Economic Modelling*, 81:242–273.
- Albonico, A., Kalyvitis, S., and Pappa, E. (2014). Capital maintenance and depreciation over the business cycle. Journal of Economic Dynamics and Control, 39:273–286.
- Albonico, A., Paccagnini, A., and Tirelli, P. (2019b). Limited asset market participation and the Euro Area crisis: An empirical DSGE model. *Economic Inquiry*, 57(3):1302– 1323.
- Amir-Ahmadi, P., Cortes, G. S., and Weidenmier, M. D. (2020). Regional monetary policies and the Great Depression. Working Paper 26695, National Bureau of Economic Research.
- Bartocci, A., Notarpietro, A., and Pisani, M. (2020). The COVID-19 shock and a fiscalmonetary policy mix in a monetary union. Temi di discussione (Economic working papers) 1313, Bank of Italy, Economic Research and International Relations Area.
- Batini, N., Cantelmo, A., Melina, G., and Villa, S. (2021). How loose, how tight? A measure of monetary and fiscal stance for the Euro Area. Oxford Economic Papers, 73(4):1536–1556.
- Batini, N., Melina, G., and Villa, S. (2019). Fiscal buffers, private debt, and recession: The good, the bad and the ugly. *Journal of Macroeconomics*, 62:103044.
- Beraja, M., Fuster, A., Hurst, E., and Vavra, J. (2019). Regional heterogeneity and the refinancing channel of monetary policy. *The Quarterly Journal of Economics*, 134(1):109–183.
- Bottone, M., Conflitti, C., Riggi, M., and Tagliabracci, A. (2021). Firms' inflation expectations and pricing strategies during Covid-19. Questioni di Economia e Finanza (Occasional Papers) 619, Bank of Italy, Economic Research and International Relations Area.
- Breuss, F. and Rabitsch, K. (2009). An estimated two-country DSGE model of Austria and the Euro Area. *Empirica*, 36(1):123–158.
- Brinca, P., Duarte, J. B., and Faria-e Castro, M. (2021). Measuring labor supply and demand shocks during COVID-19. *European Economic Review*, 139:103901.
- Cantelmo, A. and Melina, G. (2018). Monetary policy and the relative price of durable goods. *Journal of Economic Dynamics and Control*, 86:1–48.
- Cantore, C., Levine, P., and Melina, G. (2014). A fiscal stimulus and jobless recovery. *The Scandinavian Journal of Economics*, 116(3):669–701.

- Cardani, R., Croitorov, O., Giovannini, M., Pfeiffer, P., Ratto, M., and Vogel, L. (2022). The Euro Area's pandemic recession: A DSGE-based interpretation. *Journal* of Economic Dynamics and Control, 143:104512.
- Carrillo, J., Fève, P., and Matheron, J. (2018). Monetary policy inertia or persistent shocks: a DSGE analysis. *International Journal of Central Banking*, 8:1–39.
- Castelnuovo, E. and Nisticó, S. (2010). Stock market conditions and monetary policy in a DSGE model for the US. *Journal of Economic Dynamics and Control*, 34(9):1700–1731.
- Christiano, L., Eichenbaum, M., and Rebelo, S. (2011). When is the government spending multiplier large? *Journal of Political Economy*, 119(1):78–121.
- Christiano, L., Motto, R., and Rostagno, M. (2014). Risk Shocks. American Economic Review, 104(1):27–65.
- Christiano, L. J., Eichenbaum, M., and Evans, C. L. (2005). Nominal rigidities and the dynamic effects of a shock to monetary policy. *Journal of Political Economy*, 113(1):1– 45.
- Coenen, G., Karadi, P., Schmidt, S., and Warne, A. (2018). The New Area-Wide Model II: an extended version of the ECB's micro-founded model for forecasting and policy analysis with a financial sector. Working Paper Series 2200, European Central Bank.
- Corrado, L., Grassi, S., and Paolillo, A. (2021). Modelling and estimating large macroeconomic shocks during the pandemic. *CEIS Working Paper*.
- Corsetti, G., Duarte, J. B., and Mann, S. (2022). One money, many markets. *Journal of the European Economic Association*, 20(1):513–548.
- Cortes, G. S., Gao, G. P., Silva, F. B., and Song, Z. (2022). Unconventional monetary policy and disaster risk: Evidence from the subprime and COVID–19 crises. *Journal of International Money and Finance*, 122:102543.
- De Graeve, F. (2008). The external finance premium and the macroeconomy: US post-WWII evidence. Journal of Economic Dynamics and Control, 32(11):3415–3440.
- Del Negro, M., Gleich, A., Goyal, S., Johnson, A., and Tambalotti, A. (2022). Drivers of inflation: The New York Fed DSGE model's perspective. *Federal Reserve Bank of New York Liberty Street Economics*, 2022/03(1).
- Drygalla, A., Holtemoller, O., and Kiesel, K. (2020). The effects of fiscal policy in an estimated DSGE model–The case of the German stimulus packages during the Great Recession. *Macroeconomic Dynamics*, 24(6):1315–1345.
- ECB (2008). Wage indexation mechanisms in euro area contries. *ECB Monthly Bulletin*, 05.
- ECB (2023). The pandemic emergency purchase programme an initial review. *ECB* Monthly Bulletin, 08.
- Eichenbaum, M. S., Rebelo, S., and Trabandt, M. (2021). The macroeconomics of epidemics. *The Review of Financial Studies*, 34(11):5149–5187.

- Elsby, M. W., Michaels, R., and Solon, G. (2009). The ins and outs of cyclical unemployment. *American Economic Journal: Macroeconomics*, 1(1):84–110.
- Fagan, G., Henry, J., and Mestre, R. (2005). An area-wide model for the Euro Area. ECB Working Paper, 42.
- Favara, G., Gilchrist, S., Lewis, K., and Zakrajšek, E. (2016). Recession risk and the excess bond premium. *FEDS Notes*, (2016-04):15.
- Fernald, J. G. and Li, H. (2022). The impact of COVID on productivity and potential output. Technical report, Federal Reserve Bank of San Francisco.
- Forni, L., Gerali, A., and Pisani, M. (2010). The macroeconomics of fiscal consolidations in euro area countries. *Journal of Economic Dynamics and Control*, 34(9):1791–1812.
- Foroni, C., Furlanetto, F., and Lepetit, A. (2018). Labor supply factors and economic fluctuations. *International Economic Review*, 59(3):1491–1510.
- Furlanetto, F., Ravazzolo, F., and Sarferaz, S. (2019). Identification of financial factors in economic fluctuations. *The Economic Journal*, 129(617):311–337.
- Furlanetto, F. and Seneca, M. (2014). New perspectives on depreciation shocks as a source of business cycle fluctuations. *Macroeconomic Dynamics*, 18(6):1209–1233.
- Galí, J., Smets, F., and Wouters, R. (2012). Unemployment in an estimated New Keynesian model. *NBER macroeconomics annual*, 26(1):329–360.
- Gautier, E., Conflitti, C., Faber, R. P., Fabo, B., Fadejeva, L., Jouvanceau, V., Menz, J.-O., Messner, T., Petroulas, P., Roldan-Blanco, P., Rumler, F., and Se (2022). New facts on consumer price rigidity in the Euro Area. Temi di discussione (Economic working papers) 1375, Bank of Italy, Economic Research and International Relations Area.
- Gelain, P. and Ilbas, P. (2017). Monetary and macroprudential policies in an estimated model with financial intermediation. *Journal of Economic Dynamics and Control*, 78:164–189.
- Gerali, A., Locarno, A., Notarpietro, A., and Pisani, M. (2018). The sovereign crisis and Italy's potential output. *Journal of Policy Modeling*, 40(2):418–433.
- Gerali, A., Neri, S., Sessa, L., and Signoretti, F. (2010). Credit and banking in a DSGE model of the Euro Area. *Journal of Money, Credit and Banking*, 42:107–141.
- Gertler, M. and Karadi, P. (2011). A model of unconventional monetary policy. *Journal* of Monetary Economics, 58(1):17–34.
- Giagheddu, M. and Papetti, A. (2023). The macroeconomics of age-varying epidemics. *European Economic Review*, 151:104346.
- Gilchrist, S. and Mojon, B. (2018). Credit risk in the Euro Area. *The Economic Journal*, 128(608):118–158.
- Gilchrist, S. and Zakrajšek, E. (2012). Credit spreads and business cycle fluctuations. American Economic Review, 102(4):1692–1720.

- Giovannini, M., Hohberger, S., Kollmann, R., Ratto, M., Roeger, W., and Vogel, L. (2019). Euro Area and US external adjustment: The role of commodity prices and emerging market shocks. *Journal of International Money and Finance*, 94:183–205.
- Gnocchi, S., Lagerborg, A., and Pappa, E. (2015). Do labor market institutions matter for business cycles? *Journal of Economic Dynamics and Control*, 51:299–317.
- Groth, C. and Khan, H. (2010). Investment adjustment costs: An empirical assessment. Journal of Money, Credit and Banking, 42(8):1469–1494.
- Harding, M., Lindé, J., and Trabandt, M. (2022). Resolving the missing deflation puzzle. Journal of Monetary Economics, 126:15–34.
- Hirakata, N., Sudo, N., and Ueda, K. (2013). Capital injection, monetary policy, and financial accelerators. *International Journal of Central Banking*, 9(2):101–145.
- Iacoviello, M. (2015). Financial business cycles. Review of Economic Dynamics, 18(1):140–163.
- Justiniano, A., Primiceri, G., and Tambalotti, A. (2010). Investment shocks and business cycles. *Journal of Monetary Economics*, 57(2):132–145.
- Justiniano, A., Primiceri, G. E., and Tambalotti, A. (2013). Is there a trade-off between inflation and output stabilization? *American Economic Journal: Macroeconomics*, 5(2):1–31.
- Kaplan, G., Moll, B., and Violante, G. L. (2020). The Great Lockdown and the big stimulus: Tracing the pandemic possibility frontier for the U.S. Working Paper 27794, National Bureau of Economic Research.
- Kara, E. and Pirzada, A. (2020). A possible explanation of the missing deflation puzzle. *Economic Inquiry*, 58(1):361–373.
- Kiley, M. T. and Sim, J. W. (2014). Bank capital and the macroeconomy: Policy considerations. *Journal of Economic Dynamics and Control*, 43:175–198.
- Kollmann, R. (2021). Effects of Covid-19 on Euro Area GDP and inflation: Demand vs. supply disturbances. *International Economics and Economic Policy*, 18(3):475–492.
- Kollmann, R., Pataracchia, B., Raciborski, R., Ratto, M., Roeger, W., and Vogel, L. (2016). The post-crisis slump in the Euro Area and the US: Evidence from an estimated three-region DSGE model. *European Economic Review*, 88:21–41.
- Kollmann, R., Ratto, M., Roeger, W., and in't Veld, J. (2013). Fiscal policy, banks and the financial crisis. *Journal of Economic Dynamics and Control*, 37(2):387–403.
- Kollmann, R., Ratto, M., Roeger, W., in't Veld, J., and Vogel, L. (2015). What drives the German current account? And how does it affect other EU Member States? *Economic Policy*, 30(81):47–93.
- Le, V. P. M., Meenagh, D., and Minford, P. (2021). State-dependent pricing turns money into a two-edged sword: A new role for monetary policy. *Journal of International Money and Finance*, 119:102496.
- Lenza, M. and Primiceri, G. E. (2022). How to estimate a vector autoregression after March 2020. Journal of Applied Econometrics, 37(4):688–699.
- Lim, G. and McNelis, P. D. (2016). Quasi-monetary and quasi-fiscal policy rules at the zero-lower bound. *Journal of International Money and Finance*, 69:135–150.
- Lubik, T. and Schorfheide, F. (2005). A Bayesian look at new open economy macroeconomics. *NBER Macroeconomics Annual*, 20:313–366.
- Meenagh, D., Minford, P., and Wickens, M. R. (2022). The macroeconomic controversy over price rigidity-how to resolve it and how Bayesian estimation has led us astray. *Open Economies Review*, 33:617–630.
- Meh, C. A. and Moran, K. (2010). The role of bank capital in the propagation of shocks. Journal of Economic Dynamics and Control, 34(3):555–576.
- Melina, G. and Villa, S. (2014). Fiscal policy and lending relationships. *Economic Inquiry*, 52(2):696–712.
- Mouabbi, S. and Sahuc, J.-G. (2019). Evaluating the macroeconomic effects of the ECB's unconventional monetary policies. *Journal of Money, Credit and Banking*, 51(4):831–858.
- Nakamura, E. and Steinsson, J. (2008). Five facts about prices: A reevaluation of menu cost models. *The Quarterly Journal of Economics*, 123(4):1415–1464.
- Nunziata, L. (2003). Labour market institutions and the cyclical dynamics of employment. Labour Economics, 10(1):31–53.
- Ohanian, L. E. and Raffo, A. (2012). Aggregate hours worked in OECD countries: New measurement and implications for business cycles. *Journal of Monetary Economics*, 59(1):40–56.
- Poutineau, J.-C. and Vermandel, G. (2015). Cross-border banking flows spillovers in the Eurozone: Evidence from an estimated DSGE model. *Journal of Economic Dynamics* and Control, 51:378–403.
- Quint, D. and Tristani, O. (2018). Liquidity provision as a monetary policy tool: The ECB's non-standard measures after the financial crisis. *Journal of International Money* and Finance, 80:15–34.
- Ramey, V. A. and Zubairy, S. (2018). Government spending multipliers in good times and in bad: Evidence from US historical data. *Journal of Political Economy*, 126(2):850– 901.
- Riggi, M. (2019). Capital destruction, jobless recoveries, and the discipline device role of unemployment. *Macroeconomic Dynamics*, 23(2):590–624.
- Sahuc, J.-G. (2016). The ECB's asset purchase programme: A model-based evaluation. *Economics Letters*, 145:136–140.
- Santacreu, A. M. (2016). Employment and capacity utilization over the business cycle. *Economic Synopses*, 19.

- Silva, F. B. G. (2021). Fiscal deficits, bank credit risk, and loan-loss provisions. *Journal* of Financial and Quantitative Analysis, 56(5):1537–1589.
- Smets, F. and Villa, S. (2016). Slow recoveries: Any role for corporate leverage? Journal of Economic Dynamics and Control, 70:54–85.
- Smets, F. and Wouters, R. (2003). An estimated dynamic stochastic general equilibrium model of the Euro Area. Journal of the European Economic Association, 1(5):1123– 1175.
- Smets, F. and Wouters, R. (2005). Comparing shocks and frictions in US and Euro Area business cycles: A Bayesian DSGE approach. *Journal of Applied Econometrics*, 20(2):161–183.
- Smets, F. and Wouters, R. (2007). Shocks and frictions in US business cycles: A Bayesian DSGE approach. American Economic Review, 97(3):586–606.
- Spiegel, M. M. (2022). Monetary policy spillovers under COVID-19: Evidence from lending by US foreign bank subsidiaries. *Journal of International Money and Finance*, 122:102550.
- Taylor, J. B. and Wieland, V. (2012). Surprising comparative properties of monetary models: results from a new model database. *Review of Economics and Statistics*, 94(3):800–816.
- Villa, S. (2016). Financial frictions in the Euro Area and the United States: A Bayesian assessment. *Macroeconomic Dynamics*, 20(5):1313–1340.
- Woodford, M. (2003). Interest and Prices: Foundations of a Theory of Monetary Policy. Princeton University Press, Princeton, NJ.
- Wu, J. C. and Xia, F. D. (2016). Measuring the macroeconomic impact of monetary policy at the zero lower bound. *Journal of Money, Credit and Banking*, 48(2-3):253–291.
- Wu, J. C. and Xia, F. D. (2017). Time-varying lower bound of interest rates in Europe. *Chicago Booth Research Paper*, (17-06).
- Wu, J. C. and Xia, F. D. (2020). Negative interest rate policy and the yield curve. *Journal* of Applied Econometrics, 35(6):653–672.

Appendix

A Data

This section discusses the sources and transformation of the ten variables used in the estimation. Following Smets and Wouters (2007), GDP, consumption and investment are transformed in per-capita terms by dividing their real values by the labor force. Real wages are computed by dividing compensation per hour by the GDP deflator. As shown in the measurement equations in Subsection 3.1, the observable variables of GDP, consumption, investment and wages are logged and expressed in first differences. Capital utilization is computed as the first difference of the log of the rate of capacity utilization. The corresponding code is BSCURT02EZQ160S for the EA (computed from the business tendency surveys for manufacturing) and TCU for the US economy, both extracted from the ALFRED database.²²

EA data on hours worked come from Ohanian and Raffo (2012) until 2016Q4 for the following available countries: Austria, Finland, France, Germany, Ireland, Italy, and Spain. The EA variable is created by aggregating countries series weighted by countries' nominal GDP. From 2017Q1 onward EA data are extracted from Eurostat, code lfsi_ahw_q. US data are computed as average weekly hours in the nonfarm business sector (PRS85006023).

EA real net worth is computed as the difference between loans and deposits of households and corporates, divided by the GDP deflator. EA credit spread is computed as the spread of non-financial corporates with respect to domestic sovereign bonds.²³ US real net worth is given by the difference between bank credit (TOTBKCR) and deposits (DPSACBW027SBOG) of all commercial banks, divided by the GDP deflator. The credit spread is computed as in Gilchrist and Zakrajšek (2012) because, as noted by Gelain and Ilbas (2017), this is closely related to measures of financial intermediary health, which makes the spread a good predictor of distress in the financial intermediation sector.

B Model Dynamics

Figures B.1 and B.2 report posterior impulse responses of output, inflation and the monetary policy rate to all shocks in the estimated models for the EA and the US economy, respectively. All shocks are of size equal to their estimated standard deviations

²²Both measures refer to the manufacturing sector. Capital utilization rates in the service sector are generally measured less precisely and long time series comparable between the eruo area and the United States are not available.

 $^{^{23} \}rm Data$ on balance sheets and on credit spreads are available at https://www.euro-area-statistics.org/ and https://publications.banque-france.fr/en/economic-and-financial-publications-working-papers/credit-risk-euro-area, respectively.



Figure B.1: Posterior Impulse Response Functions to All Shocks in the Euro Area (All Shocks Are Set to Produce a Downturn)

and set to display an output contraction in all cases. Demand and supply shocks behave in the expected manner as far as the sign of the inflation response is concerned.



Figure B.2: Posterior Impulse Response Functions to All Shocks in the United States (All Shocks Are Set to Produce a Downturn)

C Robustness checks

C.1 Estimation Excluding the War in Ukraine

Given that the sample spans from 2000Q1 to 2022Q1, it includes the start of the war in Ukraine. This war triggered first-moment (negative) and second-moment (increasing uncertainty) shocks globally, especially in commodity markets, which affected capital utilization and inflation. As a robustness check, the model is re-estimated excluding the last quarter of the sample. Table C.1 shows that posterior parameter estimates are virtually the same as those obtained using the baseline sample.

		Prio	r distribut	ion	Posterior mean				
Parameters		Distr	Mean	Std./df	Euro area	United States			
Structural parameters				,					
Habit parameter	h	Beta	0.7	0.1	0.13 [0.10;0.15]	0.12 [0.10; 0.13]			
Inv. of Frisch elasticity	ϕ	Gamma	0.33	0.25	0.22 [0.20;0.24]	0.39 [0.15; 0.61]			
Calvo prices	σ_p	Beta	0.5	0.05	0.76 [0.72;0.79]	0.72 [0.67;0.77]			
Calvo wages	σ_w	Beta	0.5	0.05	0.31 [0.28;0.34]	0.78 [0.65;0.92]			
Price indexation	σ_{pi}	Beta	0.5	0.15	0.15 [0.05;0.24]	0.57 [0.50;0.64]			
Wage indexation	σ_{wi}	Beta	0.5	0.15	0.49 [0.29;0.70]	0.67 [0.47;0.87]			
Inv. adj. costs	ξ	Normal	4	0.5	3.20 [3.00;3.41]	3.63 [3.15;4.10]			
Elasticity of capital util	ζ	Beta	0.25	0.15	0.61 [0.52;0.69]	0.63 [0.56;0.71]			
Inflation - Taylor rule	ρ_{π}	Normal	1.7	0.15	2.03 [1.83;2.22]	1.78 [1.59;1.98]			
Output - Taylor rule	ρ_y	Gamma	0.125	0.05	0.02 [0.01;0.03]	0.11 [0.07;0.14]			
Taylor rule changes in y	ρ_{Δ_y}	Normal	0.0625	0.05	0.12 [0.09;0.16]	0.08 [0.05;0.11]			
Taylor rule smoothing	ρ_i	Beta	0.75	0.1	0.65 [0.58;0.72]	0.76 [0.72;0.81]			
Constants	1.0					- · · [- · ,- ·]			
Trend	$\bar{\gamma}$	Normal	0.4	0.2	0.02 [0.00;0.05]	0.28 [0.25; 0.32]			
Inflation	π	Gamma	0.5	0.1	0.57 [0.47;0.67]	0.68 [0.50;0.86]			
Interest rate	\bar{R}	Normal	0.8	0.2	0.30 [0.12;0.48]	0.35 [0.16; 0.55]			
Hours	$\bar{\ell}$	Normal	0.0	2.0	-0.39 [-2.43;1.71]	3.34 [1.86;5.00]			
Utilization rate	\overline{U}	Normal	0.0	0.2	-0.27 [-0.30;-0.23]	0.00 [0.03;0.03]			
Spread	\bar{S}	Gamma	0.5	0.1	0.37 [0.27;0.46]	$0.44 \ [0.36; 0.52]$			
Net worth over GDP	\bar{NY}	Normal	0.05	0.2	0.06 [-0.26;0.38]	0.06 [-0.26;0.36]			
Exogenous processes	1,1	rtorinar	0.00	0.2		0.00 [0.20,0.00]			
Technology	$ ho_a$	Beta	0.5	0.2	0.93 [0.98;0.99]	0.86 [0.80;0.93]			
reemonogy	σ_a	IG	0.1	0.2 2	0.81 [0.70;0.91]	$0.73 \ [0.63; 0.83]$			
Price mark-up		Beta	0.5	0.2	0.99 [0.98;1.00]	$0.91 \ [0.87; 0.94]$			
Thee mark up	$ ho_p \ \sigma_p$	IG	0.1	0.2 2	0.28 [0.22;0.33]	$0.24 \ [0.19; 0.29]$			
	-	Beta	0.1	0.2	$0.20 \ [0.22, 0.00]$ $0.50 \ [0.38; 0.62]$	0.24 [0.13; 0.23] 0.69 [0.58; 0.81]			
Labor	μ_p	Beta	0.5	0.2 0.2	0.97 [0.94;0.99]	0.98 [0.96; 0.99]			
Labor	$ ho_w$	IG	0.0	0.2 2	$0.30 \ [0.24; 0.35]$	0.58 [0.50; 0.59] 0.72 [0.59; 0.86]			
	σ_w	Beta	$0.1 \\ 0.5$	0.2	$0.30 \ [0.24; 0.35]$ $0.18 \ [0.05; 0.31]$	0.72 [0.55; 0.85] 0.75 [0.65; 0.85]			
Capital utilization	μ_w	Beta	0.5	$0.2 \\ 0.2$	$\begin{array}{c} 0.18 \ [0.03, 0.31] \\ 0.92 \ [2.38, 3.43] \end{array}$	0.73 [0.03; 0.83] 0.83 [0.76; 0.91]			
Capital utilization	$ ho_u$	IG	0.3 0.1	$\frac{0.2}{2}$	$\begin{array}{c} 0.32 \ [2.38, 3.43] \\ 2.90 \ [0.94; 0.98] \end{array}$	1.83 [1.56; 2.10]			
Capital quality	σ_u	Beta	$0.1 \\ 0.5$	0.2	0.96 [0.94;0.98]	$0.73 \ [0.67; 0.79]$			
Capital quality	$ ho_{kq}$	IG	0.3 0.1	$\frac{0.2}{2}$	0.30 [0.34, 0.38] 0.12 [0.09; 0.15]	0.13 [0.07; 0.13] 0.37 [0.28; 0.46]			
Inv. specific	σ_{kq}	Beta	$0.1 \\ 0.5$	0.2	$0.12 \ [0.09; 0.13]$ $0.18 \ [0.09; 0.28]$	0.37 [0.23; 0.40] 0.40 [0.29; 0.52]			
mv. specific	$ ho_x$	IG	0.5 0.1	$\frac{0.2}{2}$	7.55 [6.26;8.80]	$\begin{array}{c} 0.40 \ [0.29; 0.32] \\ 4.86 \ [3.78; 5.94] \end{array}$			
Duefenence	σ_x			$\frac{2}{0.2}$. ,				
Preference	$ ho_b$	Beta IG	$\begin{array}{c} 0.5 \\ 0.1 \end{array}$	0.2 2	$\begin{array}{c} 0.78 \ [0.70; 0.87] \\ 1.10 \ [0.88; 1.31] \end{array}$	$\begin{array}{c} 0.99 \ [0.99; 0.99] \\ 3.29 \ [2.78; 3.84] \end{array}$			
Monetary policy	σ_b	IG Beta	$0.1 \\ 0.5$	2 0.2	$\begin{array}{c} 1.10 \ [0.88; 1.31] \\ 0.38 \ [0.29; 0.47] \end{array}$	3.29 [2.78; 3.84] 0.44 [0.37; 0.52]			
monetary policy	$ ho_m$. ,			
Community on the second second	σ_m	IG Data	0.1	2	0.27 [0.22; 0.31]	0.19 [0.16; 0.22]			
Government spending	$ ho_g$	Beta	0.5	0.2	0.63 [0.48;0.77]	0.91 [0.87;0.96]			
Ct J Mar	σ_g	IG IC	0.1	2	1.90 [1.66; 2.14]	2.83 [2.47;3.17]			
Std- Measurement error		IG	0.1	2	0.15 [0.13;0.17]	0.15 [0.13; 0.17]			

Table C.1: Prior and Posterior Distributions of Structural Parameters, Constants andExogenous Processes-Excluding 2022Q1 (95 percent credible intervals in square brackets)

Figure C.1: Historical Shock Decomposition of Quarter-on-Quarter GDP Growth (Demeaned)–Excluding 2022Q1



(a) Euro Area GFC



Figures C.1, C.2, C.3 and C.4 show that the results of the paper are very robust to the exclusion of the last quarter of the sample.

Figure C.2: Historical Shock Decomposition of Quarter-on-Quarter Inflation Rate (Demeaned)–Excluding 2022Q1



(a) Euro Area GFC



Figure C.3: Contribution of Monetary Policy Shocks to Quarter-on-Quarter GDP Growth (Demeaned)–Excluding 2022Q1



(b) United States



Figure C.4: Contribution of Monetary Policy Shocks to Quarter-on-Quarter Inflation (Demeaned)–Excluding 2022Q1



C.2 Estimation with Alternative Priors

The posterior estimates of the habit formation (EA and US economy), wage and price stickiness (EA and US economy), and the responsiveness of the nominal interest rate to the output gap (EA) are close to the extremes of the prior distributions. As noted by Meenagh et al. (2022), the inappropriate choice of priors could bias Bayesian estimates. We then conduct a robustness exercise by changing the prior mean of habit parameter to 0.25 (from 0.7), the prior standard deviations of the price and wage stickiness and of the Taylor rule parameter to 0.1 (from 0.05).

Table C.2 shows that parameter estimates both of the above-mentioned parameters and all the other structural parameters are similar to the baseline estimation. The same result applies to the estimated exogenous processes. Although there are small differences, the mean estimates of the parameters under this alternative specification fall in the estimated confidence band for the same parameter of the baseline model (Table 2). This can be interpreted as a rough measure of similarity.

Consequently, Figures C.5, C.6, C.7 and C.8 show that the results of the paper are robust to the use of the alternative priors.

		Prior distri			ution		Posterior mean			
Parameters		Distr	Mean		Std./df		Euro area		United States	
			Base	Alter.	Base Alter.		Base	Alter.	Base	Alter.
Structural parameters										
Habit parameter	h	Beta	0.7 0.25		0.	.1	0.13	0.11	0.12	0.10
Inv. of Frisch elasticity	ϕ	Gamma	0.33		0.25		0.22	0.21	0.41	0.41
Calvo prices	σ_p	Beta	0.5		0.05	0.1	0.75	0.78	0.71	0.77
Calvo wages	σ_w	Beta	0.5		0.05	0.1	0.31	0.30	0.78	0.72
Price indexation	σ_{pi}	Beta	0.5		0.15		0.16	0.15	0.57	0.59
Wage indexation	σ_{wi}	Beta	0.5		0.15		0.47	0.47	0.67	0.67
Inv. adj. costs	ξ	Normal	4		0.5		3.21	3.12	3.54	3.64
Elasticity of capital util	ζ	Beta	0	0.25		0.15		0.62	0.64	0.63
Inflation - Taylor rule	ρ_{π}	Normal	1.7		0.15		2.05	2.05	1.78	1.79
Output - Taylor rule	$ ho_y$	Gamma	0.125		0.05 0.1		0.02	0.02	0.11	0.11
Taylor rule changes in y	ρ_{Δ_y}	Normal	0.0625		0.05		0.11	0.11	0.08	0.08
Taylor rule smoothing	$ ho_i$	Beta	0.75		0.1		0.67	0.67	0.76	0.76
Constants										
Trend	$\bar{\gamma}$	Normal	(0.4		0.2		0.02	0.29	0.28
Inflation	$\bar{\pi}$	Gamma	0.5		0.1		0.58	0.58	0.66	0.66
Interest rate	\bar{R}	Normal	0.8		0.2		0.31	0.31	0.33	0.33
Hours	$\bar{\ell}$	Normal	0.0		2.0		-0.12	-0.16	3.45	3.53
Utilization rate	\bar{U}	Normal	().0	0.2		-0.27	-0.27	0.01	0.00
Spread	\bar{S}	Gamma	(0.5		0.1		0.37	0.44	0.43
Net worth over GDP	\bar{NY}	Normal	0.05		0.2		0.08	0.04	0.06	0.03
Exogenous processes										
Technology	$ ho_a$	Beta	0.5		0.2		0.93	0.93	0.84	0.86
	σ_a	IG	().1	2	2	0.85	0.84	0.75	0.74
Price mark-up	$ ho_p$	Beta	0.5		0.2		0.99	0.99	0.91	0.90
	σ_p	IG	0.1		2		0.28	0.25	0.24	0.22
	μ_p	Beta	().5	0.	.2	0.51	0.54	0.69	0.69
Labor	ρ_w	Beta	0.5		0.2		0.97	0.97	0.98	0.95
	σ_w	IG	0.1		2		0.30	0.31	0.72	0.59
	μ_w	Beta	().5	0.	.2	0.18	0.18	0.75	0.81
Capital utilization	$ ho_u$	Beta	().5	0.	.2	0.92	0.93	0.84	0.84
	σ_u	IG	().1	2	2	2.82	2.80	1.83	1.82
Capital quality	ρ_{kq}	Beta	().5	0.	.2	0.96	0.96	0.74	0.72
	σ_{kq}	IG	().1		2	0.12	0.12	0.36	0.37
Inv. specific	ρ_x	Beta	0.5		0.2		0.19	0.19	0.39	0.39
	σ_x	IG	().1	2	2	7.50	7.38	4.71	4.74
Preference	$ ho_b$	Beta	0.5		0.2		0.77	0.80	0.99	0.99
σ_b		IG	0.1		2		1.12	1.09	3.28	3.32
Monetary policy	$ ho_m$	Beta	0.5		0.2		0.35	0.35	0.45	0.44
· _ •	σ_m	IG	().1	2	2	0.27	0.27	0.19	0.18
Government spending	ρ_g	Beta	0.5		0.2		0.63	0.63	0.91	0.91
-	σ_g	IG	().1		2	1.94	1.93	2.94	2.92
Std- Measurement error	2	IG	0.1		2		0.15	0.15	0.15	0.15

 Table C.2: Distributions of Structural Parameters, Constants and Exogenous Processes–

 Alternative Priors

Figure C.5: Historical Shock Decomposition of Quarter-on-Quarter GDP Growth (Demeaned)–Alternative Priors









Figure C.6: Historical Shock Decomposition of Quarter-on-Quarter Inflation Rate (Demeaned)–Alternative Priors



(a) Euro Area GFC



Figure C.7: Contribution of Monetary Policy Shocks to Quarter-on-Quarter GDP Growth (Demeaned)–Alternative Priors



(b) United States



Figure C.8: Contribution of Monetary Policy Shocks to Quarter-on-Quarter Inflation (Demeaned)–Alternative Priors



(a) Euro Area

(b) United States



RECENTLY PUBLISHED "TEMI" (*)

- N. 1399 *Measuring peer effects in parental leaves: evidence from a reform*, by Davide Dottori, Francesca Modena and Giulia Martina Tanzi (February 2023).
- N. 1400 Firms' innovation and university cooperation. New evidence from a survey of Italian firms, by Daniela Bragoli, Flavia Cortelezzi and Massimiliano Rigon (February 2023).
- N. 1401 Schools and the transmission of Sars-Cov-2: evidence from Italy, by Salvatore Lattanzio (February 2023).
- N. 1402 *Remittances in times of crisis: evidence from Italian corridors*, by Alessio Ciarlone (February 2023).
- N. 1403 *EU banks' dividend policies: main determinants and the role of capital ratios*, by Salvatore Cardillo and Jacopo Raponi (February 2023).
- N. 1404 Financial fragilities and risk-taking of corporate bond funds in the aftermath of central bank policy interventions, by Nicola Branzoli, Raffaele Gallo, Antonio Ilari and Dario Portioli (March 2023).
- N. 1405 Forecasting fiscal crises in emerging markets and low-income countries with machine learning models, by Raffaele De Marchi and Alessandro Moro (March 2023).
- N. 1406 The role of banks' technology adoption in credit markets during the pandemic, by Nicola Branzoli, Edoardo Rainone and Ilaria Supino (March 2023).
- N. 1407 The macroeconomic effects of temperature surprise shocks, by Filippo Natoli (March 2023).
- N. 1408 *The impact of credit substitution between banks on investment*, by Francesco Bripi (April 2023).
- N. 1409 EU structural funds and GDP per capita: spatial var evidence for the European regions, by Sergio Destefanis and Valter Di Giacinto (April 2023).
- N. 1410 Should inequality factor into central banks' decisions?, by Niels-Jakob H. Hansen, Alessandro Lin and Rui C. Mano (April 2023).
- N. 1411 The anatomy of labor cost adjustment to demand shocks: Germany and Italy during the Great Recession, by Francesco D'Amuri, Salvatore Lattanzio and Benjamin S. Smith (June 2023).
- N. 1412 *Quantitative easing, accounting and prudential frameworks, and bank lending,* by Andrea Orame, Rodney Ramcharan and Roberto Robatto (June 2023).
- N. 1413 How costly are cartels?, by Flavien Moreau and Ludovic Panon (June 2023).
- N. 1414 Trade in services related to intangibles and the profit shifting hypothesis, by Nadia Accoto, Stefano Federico and Giacomo Oddo (June 2023).
- N. 1415 *Currency risk premiums redux*, by Federico Calogero Nucera, Lucio Sarno and Gabriele Zinna (July 2023).
- N. 1416 *The external financial spillovers of CBDCs*, by Alessandro Moro and Valerio Nispi Landi (July 2023).
- N. 1417 Parental retirement and fertility decisions across family policy regimes, by Edoardo Frattola (July 2023).
- N. 1418 *Entry, exit, and market structure in a changing climate*, by Michele Cascarano, Filippo Natoli and Andrea Petrella (July 2023).
- N. 1419 Temperatures and search: evidence from the housing market, by Michele Cascarano and Filippo Natoli (July 2023).
- N. 1420 Flight to climatic safety: local natural disasters and global portfolio flows, by Fabrizio Ferriani, Andrea Gazzani and Filippo Natoli (July 2023).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ACCETTURO A., A. LAMORGESE, S. MOCETTI and D. PELLEGRINO, *Housing Price elasticity and growth: evidence from Italian cities,* Journal of Economic Geography, v. 21, 3, pp. 367-396, WP 1267 (March 2020).
- AFFINITO M. and M. PIAZZA, Always look on the bright side? Central counterparties and interbank markets during the financial crisis, International Journal of Central Banking, v. 17, 1, pp. 231-283, WP 1181 (July 2018).
- ALBANESE G., E. CIANI and G. DE BLASIO, *Anything new in town? The local effects of urban regeneration policies in Italy*, Regional Science and Urban Economics, v. 86, **WP 1214 (April 2019).**
- ALBANESE G., G. DE BLASIO and A. LOCATELLI, Does EU regional policy promote local TFP growth? Evidence from the Italian Mezzogiorno, Papers in Regional Science, v. 100, 2, pp. 327-348, WP 1253 (December 2019).
- ALBERTAZZI A., A. NOBILI and F. M. SIGNORETTI, *The bank lending channel of conventional and unconventional monetary policy*, Journal of Money, Credit and Banking, v. 53, 2-3, pp. 261-299, **WP 1094 (Jan 2017)**.
- ANZUINI A. and L. ROSSI, Fiscal policy in the US: a new measure of uncertainty and its effects on the American economy, Empirical Economics, v. 61, 6, pp. 2613-2634, WP 1197 (November 2018).
- APRIGLIANO V. and D. LIBERATI, Using credit variables to date business cycle and to estimate the probabilities of recession in real time, The Manchester School, v. 89, 51, pp. 76-96, WP 1229 (July 2019).
- AUER S., M. BERNARDINI and M. CECIONI, *Corporate leverage and monetary policy effectiveness in the euro area,* European Economic Review, v. 140, Article 103943, **WP 1258 (December 2019).**
- BANERJEE R, L. GAMBACORTA and E. SETTE, *The real effects of relationship lending*, Journal of Financial Intermediation, v. 48, Article 100923, **WP 1133 (September 2017).**
- BAMIEH O and A. CINTOLESI, *Intergenerational Transmission in regulated professions and the role of familism,* Journal of Economic Behavior & Organization, v. 192, pp. 857-879, **WP 1350 (October 2021).**
- BARONE G., F. DAVID, G. DE BLASIO and S. MOCETTI, *How do house prices respond to mortgage supply?*, Journal of Economic Geography, v. 21, 1, pp.127-140, **WP 1282 (June 2020).**
- BARONE G. and S. MOCETTI, *Intergenerational mobility in the very long run: Florence 1427-2011*, Review of Economic Studies, v. 88, 4, pp. 1863–1891, **WP 1060 (April 2016).**
- BARTOCCI A., L. BURLON, A. NOTARPIETRO and M. PISANI, *Macroeconomic effects of non-standard monetary* policy measures in the Euro Area: the role of corporate bond purchases, The Manchester School, v. 89, S1, pp. 97-130, WP 1241 (October 2019).
- BATINI N., A. CANTELMO, G. MELINA and S. VILLA, *How loose, how tight? A measure of monetary and fiscal stance for the euro area,* Oxford Economic Papers, v. 73, 4, pp. 1536-1556, **WP 1295 (September 2020).**
- BENETTON M. and D. FANTINO, *Targeted monetary policy and bank lending behavior*, Journal of Financial Economics, v. 142, 1, pp. 404-429, **WP 1187 (September 2018).**
- BUSETTI F., M. CAIVANO and D. DELLE MONACHE, Domestic and global determinants of inflation: evidence from expectile regression, Oxford Bulletin of Economics and Statistics, v. 83, 4, pp. 982-1001, WP 1225 (June 2019).
- BUSETTI F., M. CAIVANO, D. DELLE MONACHE and C. PACELLA, *The time-varying risk of Italian GDP*, Economic Modelling, v. 101, Article 105522, **WP 1288 (July 2020).**
- BUSETTI F., S. NERI, A. NOTARPIETRO and M. PISANI, Monetary Policy strategies in the new normal: a modelbased analysis for the Euro Area, Journal of Macroeconomics, v. 70, Article 103366, WP 1308 (December 2020).
- CAPOLONGO A. and C. PACELLA, Forecasting inflation in the Euro Area: countries matter, Empirical Economics, v. 61, 4, pp. 2477-2499, WP 1224 (June 2019).
- CARMIGNANI A., G. DE BLASIO, C. DEMMA and A. D'IGNAZIO, *Urbanization and firm access to credit,* Journal of Regional Science, v. 61, 3, pp. 597-622, WP 1222 (June 2019).
- CORNELI F., *Financial integration without financial development*, Atlantic Economic Journal, v. 49, 2, pp. 201-220, **WP 1120 (June 2017).**
- COVA P., P. PAGANO, A. NOTARPIETRO and M. PISANI, Secular stagnation, R&D, public investment and monetary policy: a global-model perspective, Macroeconomic Dynamics, v. 25, 5, pp. 1267-1287, WP 1156 (December 2017).
- DE PHILIPPIS M., *Multitask agents and incentives: the case of teaching and research for university professors,* Economic Journal, v. 131, 636, pp. 1643-1681, **WP 1042 (December 2015).**
- DEL PRETE S. and M. L. STEFANI, *Women as "Gold Dust": gender diversity in top boards and the performanceof Italian banks*, Economic Notes, Monte dei Paschi di Siena, v. 50, 2, e12183, WP 1014 (June 2015).

- FERRERO G., M. LOBERTO and M. MICCOLI, *The assets' pledgeability channel of unconventional monetary policy*, Economic Inquiry, v. 59, 4, pp. 1547-1568, **WP 1119 (June 2017).**
- FIDORA M., C. GIORDANO and M. SCHMITZ, *Real exchange rate misalignments in the Euro Area*, Open Economies Review, v. 32, 1, pp. 71-107, WP 1162 (January 2018).
- GAMBACORTA L., G. RICOTTI, S. SUNDARESAN and Z. WANG, *Tax effects on bank liability structure*, European Economic Review, v. 138, Article 103820, **WP 1101 (February 2017).**
- HERTWECK M., V. LEWIS and S. VILLA, *Going the extra mile: effort by workers and job-seekers,* Journal of Money, Credit and Banking, v. 54, 8, pp. 2099-2127, WP 1277 (June 2020).
- LI F., A. MERCATANTI, T. MAKINEN and A. SILVESTRINI, *A regression discontinuity design for ordinal running variables: evaluating central bank purchases of corporate bonds,* The Annals of Applied Statistics, v. 15, 1, pp. 304-322, **WP 1213 (March 2019).**
- LOSCHIAVO D., *Big-city life (dis)satisfaction? The effect of urban living on subjective well-being*, Journal of Economic Behavior & Organization, vol. 192, pp. 740-764, **WP 1221 (June 2019).**
- LOSCHIAVO D., *Household debt and income inequality: evidence from Italian survey data*, Review of Income and Wealth. v. 67, 1, pp. 61-103, **WP 1095 (January 2017).**
- METELLI L. and F. NATOLI, *The international transmission of US tax shocks: a proxy-SVAR approach*, IMF Economic Review, v. 69, 2, pp. 325-356, **WP 1223 (June 2019).**
- NISPI LANDI V. and A. SCHIAVONE, *The effectiveness of capital controls*, Open Economies Review, v. 32, 1, pp. 183-211, WP 1200 (November 2018).
- PAPETTI A., *Demographics and the natural real interest rate: historical and projected paths for the Euro Area*, Journal of Economic Dynamics and Control, v. 132, Article 04209, WP 1306 (November 2020).
- PEREDA FERNANDEZ S., Copula-based random effects models for clustered data, Journal of Business & Economic Statistics, v. 39, 2, pp. 575-588, WP 1092 (January 2017).

2022

- ANDINI M., M. BOLDRINI, E. CIANI, G. DE BLASIO, A. D'IGNAZIO and A. PALADINI, Machine learning in the service of policy targeting: the case of public credit guarantees, Journal of Economic Behavior & Organization, v. 198, pp. 434-475, WP 1206 (February 2019).
- ANGELICO C., J. MARCUCCI, M. MICCOLI and F. QUARTA, Can we measure inflation expectations using twitter?, Journal of Econometrics, v. 228, 2, pp. 259-277, WP 1318 (February 2021).
- BARTOCCI A., A. NOTARPIETRO and M. PISANI, *Covid-19 shock and fiscal-monetary policy mix in a monetary union*, Economic challenges for Europe after the pandemic, Springer Proceedings in Business and Economics, Berlin-Heidelberg, Springer, **WP 1313 (December 2020).**
- BOTTERO M., C. MINOIU, J. PEYDRÒ, A. POLO, A. PRESBITERO and E. SETTE, *Expansionary yet different: credit supply and real effects of negative interest rate policy,* Journal of Financial Economics, v. 146, 2, pp. 754-778, **WP 1269 (March 2020).**
- BRONZINI R., A. D'IGNAZIO and D. REVELLI, *Financial structure and bank relationships of Italian multinational firms*, Journal of Multinational Financial Management, v. 66, Article 100762, WP 1326 (March 2021).
- CANTELMO A., *Rare disasters, the natural interest rate and monetary policy,* Oxford Bulletin of Economics and Statistics, v. 84, 3, pp. 473-496, **WP 1309 (December 2020).**
- CARRIERO A., F. CORSELLO and M. MARCELLINO, *The global component of inflation volatility,* Journal of Applied Econometrics, v. 37, 4, pp. 700-721, **WP 1170 (May 2018).**
- CIAPANNA E. and G. ROVIGATTI, *The grocery trolley race in times of Covid-19. Evidence from Italy*, Italian Economic Journal / Rivista italiana degli economisti, v. 8, 2, pp. 471-498, **WP 1341 (June 2021)**.
- CONTI A. M., A. NOBILI and F. M. SIGNORETTI, *Bank capital requirement shocks: a narrative perspective,* European Economic Review, v.151, Article 104254, **WP 1199 (November 2018).**
- FAIELLA I. and A. MISTRETTA, *The net zero challenge for firms' competitiveness*, Environmental and Resource Economics, v. 83, pp. 85-113, **WP 1259 (February 2020).**
- FERRIANI F. and G. VERONESE, *Hedging and investment trade-offs in the U.S. oil industry*, Energy Economics, v. 106, Article 105736, WP 1211 (March 2019).
- GUISO L., A. POZZI, A. TSOY, L. GAMBACORTA and P. E. MISTRULLI, *The cost of steering in financial markets:* evidence from the mortgage market, Journal of Financial Economics, v.143, 3, pp. 1209-1226, WP 1252 (December 2019).
- LAMORGESE A. and D. PELLEGRINO, *Loss aversion in housing appraisal: evidence from Italian homeowners,* Journal of Housing Economics, v. 56, Article 101826, WP 1248 (November 2019).

- LI F., T. MÄKINEN, A. MERCATANTI and A. SILVESTRINI, *Causal analysis of central bank holdings of corporate bonds under interference*, Economic Modelling, v.113, Article 105873, WP 1300 (November 2020).
- LOBERTO M, A. LUCIANI and M. PANGALLO, *What do online listings tell us about the housing market?*, International Journal of Central Banking, v. 18, 4, pp. 325-377, **WP 1171 (April 2018).**
- MIRENDA L., M. SAURO and L. RIZZICA, *The economic effects of mafia: firm level evidence*, American Economic Review, vol. 112, 8, pp. 2748-2773, WP 1235 (October 2019).
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility,* Journal of Human Resources, v. 57, 2, pp. 525-554, WP 1182 (July 2018).
- PERICOLI M. and M. TABOGA, Nearly exact Bayesian estimation of non-linear no-arbitrage term-structure models, Journal of Financial Econometrics, v. 20, 5, pp. 807-838, WP 1189 (September 2018).
- ROSSI P. and D. SCALISE, *Financial development and growth in European regions,* Journal of Regional Science, v. 62, 2, pp. 389-411, WP 1246 (November 2019).
- SCHIVARDI F., E. SETTE and G. TABELLINI, *Credit misallocation during the European financial crisis,* Economic Journal, v. 132, 641, pp. 391-423, **WP 1139 (September 2017).**
- TABOGA M., Cross-country differences in the size of venture capital financing rounds: a machine learning approach, Empirical Economics, v. 62, 3, pp. 991-1012, WP 1243 (November 2019).

2023

- APRIGLIANO V., S. EMILIOZZI, G. GUAITOLI, A. LUCIANI, J. MARCUCCI and L. MONTEFORTE, *The power of text-based indicators in forecasting Italian economic activity*, International Journal of Forecasting, v. 39, 2, pp. 791-808, WP 1321 (March 2021).
- BARTOCCI A., A. NOTARPIETRO and M. PISANI, *Non-standard monetary policy measures in non-normal times,* International Finance, v. 26, 1, pp. 19-35, **WP 1251 (November 2019).**
- DI ADDARIO S., P. KLINE, R. SAGGIO and M. SØLVSTEN, *It ain't where you're from, it's where you're at: hiring origins, firm heterogeneity, and Wages, Journal of Econometrics, v. 233, 2, pp. 340-374, WP 1374 (June 2022).*
- FERRIANI F., *Issuing bonds during the Covid-19 pandemic: was there an ESG premium?*, International Review of Financial Analysis, v. 88, Article 102653, **WP 1392 (November 2022).**
- GIORDANO C., Revisiting the real exchange rate misalignment-economic growth nexus via the across-sector misallocation channel, Review of International Economics, v. 31, 4, pp. 1329-1384, WP 1385 (October 2022).
- GUGLIELMINETTI E., M. LOBERTO and A. MISTRETTA, *The impact of COVID-19 on the European short-term rental market*, Empirica, v. 50, 3, pp. 585-623, **WP 1379 (July 2022).**
- LILLA F., Volatility bursts: a discrete-time option model with multiple volatility components, Journal of Financial Econometrics, v. 21, 3, pp. 678-713, WP 1336 (June 2021).
- LOBERTO M., *Foreclosures and house prices*, Italian Economic Journal / Rivista italiana degli economisti, , v. 9, 1, pp. 397-424, **WP 1325 (March 2021).**
- NERI S., Long-term inflation expectations and monetary policy in the Euro Area before the pandemic, European Economic Review, v. 154, Article 104426, WP 1357 (December 2021).
- ORAME A., *Bank lending and the European debt crisis: evidence from a new survey*, International Journal of Central Banking, v. 19, 1, pp. 243-300, **WP 1279 (June 2020).**
- RIZZICA L., G. ROMA and G. ROVIGATTI, *The effects of shop opening hours deregulation: evidence from Italy,* The Journal of Law and Economics, v. 66, 1, pp. 21-52, **WP 1281 (June 2020).**
- TANZI G. M., Scars of youth non-employment and labour market conditions, Italian Economic Journal / Rivista italiana degli economisti, v. 9, 2, pp. 475-499, WP 1312 (December 2020).

FORTHCOMING

- BALTRUNAITE A., M. CANNELLA, S. MOCETTI and G. ROMA, Board composition and performance of state-owned enterprises: quasi experimental evidence, The Journal of Law, Economics, and Organization, WP 1328 (April 2021).
- CIAPANNA E, S. MOCETTI and A. NOTARPIETRO, *The macroeconomic effects of structural reforms: an empirical and model-based approach*, Economic Policy, **WP 1303 (November 2022).**
- FERRARI A. and V. NISPI LANDI, Whatever it takes to save the planet? Central banks and unconventional green policy, Macroeconomic Dynamics, WP 1320 (February 2021).

- FERRARI A. and V. NISPI LANDI, *Toward a green economy: the role of central bank's asset purchases,* International Journal of Central Banking, WP 1358 (February 2022).
- LOMBARDI M. J., M. RIGGI and E. VIVIANO, *Bargaining power and the Phillips curve: a micro-macro analysis, and wages,* Journal of the European Economic Association, **WP 1302 (November 2020).**
- MICHELANGELI V. and E. VIVIANO, *Can internet banking affect households' participation in financial markets and financial awarness?*, Journal of Money, Credit and Banking, **WP 1329 (April 2021).**
- MISTRETTA A., Synchronization vs transmission: the effect of the German slowdown on the Italian business cycle, International Journal of Central Banking, WP 1346 (October 2021).
- RAINONE E., *Real-time identification and high frequency analysis of deposits outflows*, Journal of Financial Econometrics, **WP 1319 (February 2021).**